



WORKING PAPER SERIES

NO 1690 / JULY 2014

HOUSEHOLD WEALTH IN THE EURO AREA

THE IMPORTANCE OF **INTERGENERATIONAL TRANSFERS, HOMEOWNERSHIP AND HOUSE PRICE DYNAMICS**

Thomas Y. Mathä, Alessandro Porpiglia and Michael Ziegelmeyer

> HOUSEHOLD FINANCE AND **CONSUMPTION NETWORK**



NOTE: This Working Paper should not be reported as representing the views of the European Central Bank (ECB). The views expressed are those of the authors and do not necessarily reflect those of the ECB.



Household Finance and Consumption Network

This paper contains research conducted within the Household Finance and Consumption Network (HFCN). The HFCN consists of survey specialists, statisticians and economists from the ECB, the national central banks of the Eurosystem and a number of national statistical institutes.

The HFCN is chaired by Gabriel Fagan (ECB) and Carlos Sánchez Muñoz (ECB). Michael Haliassos (Goethe University Frankfurt), Tullio Jappelli (University of Naples Federico II), Arthur Kennickell (Federal Reserve Board) and Peter Tufano (University of Oxford) act as external consultants, and Sébastien Pérez Duarte (ECB) and Jiri Slacalek (ECB) as Secretaries.

The HFCN collects household-level data on households' finances and consumption in the euro area through a harmonised survey. The HFCN aims at studying in depth the micro-level structural information on euro area households' assets and liabilities. The objectives of the network are:

1) understanding economic behaviour of individual households, developments in aggregate variables and the interactions between the two;

2) evaluating the impact of shocks, policies and institutional changes on household portfolios and other variables;

- 3) understanding the implications of heterogeneity for aggregate variables;
- 4) estimating choices of different households and their reaction to economic shocks;
- 5) building and calibrating realistic economic models incorporating heterogeneous agents;
- 6) gaining insights into issues such as monetary policy transmission and financial stability.

The refereeing process of this paper has been co-ordinated by a team composed of Gabriel Fagan (ECB), Pirmin Fessler (Oesterreichische Nationalbank), Michalis Haliassos (Goethe University Frankfurt), Tullio Jappelli (University of Naples Federico II), Sébastien PérezDuarte (ECB), Jiri Slacalek (ECB), Federica Teppa (De Nederlandsche Bank), Peter Tufano (Oxford University) and Philip Vermeulen (ECB).

The paper is released in order to make the results of HFCN research generally available, in preliminary form, to encourage comments and suggestions prior to final publication. The views expressed in the paper are the author's own and do not necessarily reflect those of the ESCB.

Acknowledgements

This paper uses data from the Eurosystem Household Finance and Consumption Survey (HFCS). The results published and the related observations and analysis may not correspond to results or analysis of the data producers. This paper should not be reported as representing the views of the BCL or the Eurosystem. The views expressed are those of the authors and may not be shared by other research staff or policymakers in the BCL or the Eurosystem. We are grateful for useful comments received from internal seminar participants at the BCL and the HFCN research seminar and participants at the Bundesbank conference "Household Finances, Saving and Inequality: An International Perspective" (March 2013) and the ECB Household Finance and Consumption conference (October 2013), as well as our discussant T. Jappelli. We would like to thank Nicole Fortin for helpful discussions and the explanations provided, and an anonymous referee, as well as D. Christelis, P. Fessler and D. Georgarakos for constructive comments.

Thomas Y. Mathä

Central Bank of Luxembourg; e-mail: thomas.mathae@bcl.lu

Alessandro Porpiglia

Central Bank of Luxembourg; e-mail: alessandro.porpiglia@bcl.lu

Michael Ziegelmeyer

Central Bank of Luxembourg and Munich Center for the Economics of Aging (MEA); e-mail: michael.ziegelmeyer@bcl.lu

© European Central Bank, 2014

Address	Kaiserstrasse 29, 60311 Frankfurt am Main, Germany
Postal address	Postfach 16 03 19, 60066 Frankfurt am Main, Germany
Telephone	+49 69 1344 0
Internet	http://www.ecb.europa.eu
Fax	+49 69 1344 6000

All rights reserved. Any reproduction, publication and reprint in the form of a different publication, whether printed or produced electronically, in whole or in part, is permitted only with the explicit written authorisation of the ECB or the authors. This paper can be downloaded without charge from http://www.ecb.europa.eu or from the Social Science Research Network electronic library at http://ssrn.com/abstract_id=2431311. Information on all of the papers published in the ECB Working Paper Series can be found on the ECB's website, http://www.ecb.europa.eu/pub/scientific/wps/date/html/index.en.html

ISSN	1725-2806 (online)
ISBN	978-92-899-1098-9
EU Catalogue No	QB-AR-14-064-EN-N (online)

Abstract:

Results from the Eurosystem Household Finance and Consumption Survey reveal substantial variation in household net wealth across euro area countries that await explanation. This paper focuses on three main factors for the wealth accumulation process, i) homeownership, ii) housing value appreciation and iii) intergenerational transfers. We show that these three factors, in addition to the common household and demographic factors, are relevant for the net wealth accumulation process in all euro area countries, and moreover that, using various decomposition techniques, differences therein, in particular in homeownership rates and house price dynamics, are important for explaining wealth differences across euro area countries.

Keywords: household wealth, homeownership, property prices, inheritance, euro area JEL Codes: D31, E21, O52, C42

Non-technical summary

Recent results published in April 2013 from the Eurosystem Household Finance and Consumption Survey reveal substantial variation in household net wealth across euro area countries that await explanation. Median household net wealth varies from \in 51,400 (Germany) to \in 397,800 (Luxembourg) with the euro area figure standing at \in 109,200. Similarly, mean household net wealth varies from \in 79,700 (Slovakia) and \in 710,100 (Luxembourg) with a euro area figure standing at \in 230,800. Thus, the natural question to ask is why are mean and median net wealth differences so large between euro area countries and what are the driving factors behind? This paper provides a first in-depth analysis of factors contributing to household wealth (differences) across euro area countries. Differences in household characteristics aside, it focuses on three major factors for the wealth accumulation process, 1) homeownership, 2) housing value appreciation and 3) intergenerational transfers that, in the household wealth literature, are repeatedly found to be of relevance for the wealth accumulation process.

First, homeownership, which is usually the most important asset in household portfolios, varies greatly in the euro area (44%-90%). Second, house price dynamics in the last 20+ years differ substantially across euro area countries. In countries with very dynamic past house price developments, households owning their dwelling benefit from accrued capital gains and the more so the earlier they invested. To ensure empirical comparability we develop a survey-based house price index that is both comparable across euro area countries and more extensive in time coverage than existing publicly available price indices. Third, intergenerational transfers increase household wealth directly through the cash value of the transfer and indirectly through their impact on the homeownership versus tenancy choice, i.e. for the decision to own or not to own.

We show not only that these three factors, in addition to the common household characteristics and demographic factors, are relevant determinants of the household net wealth accumulation process in almost every euro area country, but moreover that, using various decomposition techniques, differences therein, in particular in homeownership rates and house price dynamics, are important for explaining wealth differences across euro area countries. Across euro area countries, these factors explain on average 56% of the difference in total net household wealth at their respective median level relative to Germany. Similar results are found along the whole household net wealth distribution, although the relevance of the analysed factors in explaining household net wealth differences tend to be lower for the wealthier strata of the population.

1 Introduction

Recent empirical evidence shows that household wealth varies substantially across developed countries (e.g. Davies et al., 2011; Christelis, Georgarakos and Haliassos, 2013). Similarly, first results from the new Eurosystem Household Finance and Consumption Network (2013a) reveal substantial wealth differences across euro area countries. Figure 1 provides a graphical illustration of this. It shows the mean and median net household wealth across 15 euro area countries that participated in the first wave of the HFCS.¹ Median household net wealth in the euro area is €109,200, ranging from €51,400 (Germany) to €397,800 (Luxembourg). The corresponding mean figure for the euro area is €230,800, ranging from €79,700 (Slovakia) to €670,900 (Cyprus) and €710,100 (Luxembourg). The advantage of the new dataset is that it provides the possibility of consistent cross-country comparisons based on ex-ante harmonised questionnaires. For several euro countries, it is the first time that representative high quality household wealth data have become available



Figure 1: Mean and median net wealth in the euro area by country

Source: own calculations based on the HFCS UDB 1.0; data are multiply imputed and weighted.

Thus, the natural question to ask is why are mean and median net wealth differences so large between euro area countries and what are the driving factors behind? This paper provides a first in-depth analysis of factors contributing to household wealth (differences)

¹ If the text refers to euro area countries, it means the 15 euro area countries included in the first wave of the HFCS and excludes Estonia and Ireland.

across euro area countries. Differences in household characteristics aside, it focuses on, as we believe, three major factors for the wealth accumulation process.²

First, as it is evident from basic descriptive statistics the majority of households in the euro area are homeowners. The highest share is observed for Slovakia (90%) followed by Spain (83%) and Slovenia (82%). In two countries only the share is below 50%; Germany (44%) and Austria (48%). For most homeowners, the value of the household main residence (HMR), which is the result of initial price, accrued capital gains from increased property prices, depreciation or reinvestments, is regarded as the most valuable asset in the household wealth portfolio. The mean contribution of the HMR to total net wealth is almost 50% in the euro area. Importantly, homeowners are wealthier than their non-home owning counterparts. This applies regardless of the percentile in the net wealth distribution or the country concerned (see also Figure 2 in Section 2). Thus, household net wealth must somehow be linked to homeownership.

A priori, it is far from obvious why this should be the case, as at the time of HMR acquisition, households simply exchange financial assets for real assets (plus an eventual mortgage) whereas total net wealth stays unaffected. However, countries often promote homeownership with subsidies and tax deductible interest rate payments for mortgages etc..., which make homeownership a very attractive long-term investment relative to other financial investments, not only because it is safe, but also because it promises long-term capital gains (not least as land prices usually do not get cheaper). In some countries, homeownership is also commonly regarded as a means for old age provision, in particular if public pension rights are on the low and/or uncertain side. It is also conceivable that homeownership changes household saving and consumption behaviour. Thus, it is important to analyse how homeownership contributes to the wealth accumulation process and how differences therein can explain the household wealth differences across countries.

Second, if homeownership matters, then the dynamics of house price developments over time matter for how wealthy households are and differences therein must contribute to explaining wealth differences across countries. As we will show in this paper, the residential property prices varied indeed very substantially over time and across euro area countries, and this largely explains the observed wealth differences across countries.

² The literature on household wealth has, apart from usual household characteristics, such as income, household size, civil status, age, education, etc... shown that immigration (e.g. Bauer et al., 2011), ethnicity (e.g. Blau and Graham, 1990), intergenerational transfers (e.g. Wolff and Gittleman, 2011) play an important role for differences in the wealth accumulation of households. In a cross-country setting as our, additionally cross-country differences in institutional aspects, such as differences in fiscal measures (subsidies and taxes), provision of social housing, the regulation of the rental market, financial deregulation, banking supervision, pensions and social security come into play as contributing factors (Andrews, Caldera Sánchez and Johansson, 2011; Chiuri and Jappelli, 2003).

Third, for most households becoming a homeowner is undoubtedly related to becoming indebted with a mortgage to pay off. In many countries, obtaining a mortgage is a major hurdle, as maximum loan-to-value ratios effectively limit mortgage accessibility for households in general and young households in particular as they had less time to save for necessary down payments (e.g. Chiuri and Jappelli, 2003). In this context, the importance of intergenerational transfers for wealth accumulation becomes apparent, which is the third focus of this paper; it seeks to quantify the effect of intergenerational transfers on wealth accumulation and wealth differences in the euro area. Piketty (2011) demonstrates for example for France how important the annual flow of inheritance is for the national income and wealth accumulation process. Using the Survey of Consumer Finances from 1989-2007, Wolff and Gittleman (2011) report that 21% of U.S. households received gifts or inheritances within this period, which contributes 23% to their current net wealth. Fessler, Mooslechner and Schürz (2008) report using cross-country data from the Luxembourg Wealth Study (LWS) that households receiving an inheritance are better educated, have higher income and wealth. Our results show that intergenerational transfers (excluding inherited/gifted HMRs) contribute on average 11 percentage points to mean total net wealth. Combined with the HMR contribution of 49%, these two factors contribute 60% to mean total net wealth of households.

We proceed as follows: first, we estimate country specific and euro area median regressions. We explain the median level of total net wealth with a set of covariates, including commonly used household variables (e.g. income, age, gender, marital status, education, etc...), as well as intergenerational transfers, homeownership and house price dynamics and show that the latter three are indeed important factors for the wealth accumulation process. The second part analyses to what extent these three factors are able to explain cross-countries wealth differences. For this purpose we make use of the Oaxaca-Blinder and recentred influence function (RIF) Oaxaca Blinder decompositions. We show that net wealth differences in the euro area are to a large extent driven by cross-country differences in homeownership rates, house price dynamics and to a lesser extent in received intergenerational transfers. Across euro area countries, these factors explain on average 56% of the difference in total net household wealth at their respective median level relative to Germany. Similar results are found along the whole household net wealth distribution, although the relevance of the analysed factors in explaining household net wealth differences tend to be lower for the wealthier strata of the population.

Section 2 presents the database and introduces descriptive statistics. Section 3 presents the construction of HFCS based property price indices. Section 4 presents the estimation strategy and reports the results. In section 5 we provide additional robustness tests. Section 6 concludes.

2 Data and descriptive statistics

We use data from the Eurosystem Household Finance and Consumption Survey (HFCS). The dataset includes over 62,000 observations, which represent almost 140 million private households resident in the euro area (with exclusion of Ireland and Estonia). For a brief summary of the most pertinent facts concerning the dataset see the Appendix A. For very detailed descriptive results and methodological details, see HFCN (2013a,b). First, we show the importance and variation of the HMR ownership in the euro area and the dispersion of residential property price increases in the past decades. Second, we provide figures regarding the prevalence and size of received intergenerational transfers. Variables' definitions and detailed summary statistics are provided in Appendix B.

According to a current OECD study, homeownership has risen in many OECD countries in recent decades. Reasons may be related to demographic changes caused by ageing populations (Andrews and Caldera Sánchez, 2011). Other factors may include financial market liberalisation, for example via reduction in required down payments, allowing younger people in particular to step onto the property ladder (e.g. Chiuri and Jappelli, 2003). A robust result across countries tends to be that the ownership share varies with the demographic structure of households; for example single households are less likely to own the HMR and there is a clear trend that ownership increases with the age of the household (e.g. Chiuri and Jappelli, 2003; Andrews and Caldera Sánchez, 2011). Furthermore, there is a sizeable gap between households with no formal or primary education to all other education levels. Similarly, immigrant households are twice more likely not to own their HMR. Chiuri and Jappelli (2003) for example argue that intergenerational transfers may help households to make their down payment and thus to circumvent imperfections in mortgage markets.

Figure 2 shows mean and median net wealth of HMR owners and non-owners. A robust fact is that owners have a much higher mean and median net wealth. For example, the mean (median) net wealth of HMR owners in the euro area aggregate is \notin 351,000 (\notin 218,000) and \notin 50,000 (\notin 9,000) for non-owners. There are also large differences in HMR homeownership rates across countries (Figure 3). The average homeownership rate in the euro area is 60%. The ownership rate is lowest in Germany (44%) and highest in Slovakia (90%).

Consequently, differences in homeownership rates are partially able to explain different mean contributions of the HMR to total net wealth. The contribution is for example 38% in Germany and 74% in Slovakia (Figure 3). The mean contribution in the euro area is 49%.



Figure 2: Mean and median net wealth of HMR owners and non-owners

Source: own calculations based on the HFCS UDB 1.0; data are multiply imputed and weighted.



Figure 3: HMR ownership rate and mean contribution of the HMR to total net wealth

Figure 4 depicts the net HMR value for specific percentiles in relation to the household net wealth at the corresponding percentile. The left panel contains countries from Central and Northern Europe, the right panel Southern European countries and the panel at the bottom Eastern European countries and the euro area aggregate. In none of the countries do households own their HMR at the 10th percentile of the HMR distribution; therefore the share is zero. Moving to higher percentiles and depending on the share of homeowners in each country, we measure positive and increasing contributions of the HMR to total net wealth in various countries. For most countries, the contribution of the HMR to total net wealth decreases at higher percentiles. This is not surprising, as at the wealthier end of the net wealth distribution holding other real estate and financial assets increase in importance.

Source: own calculations based on the HFCS UDB 1.0; data are multiply imputed and weighted.



Figure 4: Contribution of the HMR to total net wealth at selected percentiles

In summary, it seems that homeownership is linked to net wealth and thus contributes to explaining the observed variation of household net wealth across euro area countries. As to the likely reasons why this is the case, we can distinguish between at least two mechanisms. First, homeownership is likely to directly affect net wealth levels in countries promoting homeownership (Andrews, Caldera Sánchez and Johansson, 2011) with tax rebates or direct and indirect subsidies.



Figure 5: Changes in residential property price indices in %, 5-year moving averages

Source: ECB Statistical Data Warehouse. Individual sources are written next to each country. The time series "new and existing dwellings" is used for all countries in the left panel, "existing dwellings" are used for all countries in the right panel.

Source: own calculations based on the HFCS UDB 1.0; data are multiply imputed and weighted.

Second, homeowners may benefit from an increase in the value of their property, either due to increased prices of their land, their dwelling or both. Residential property price indices indicate huge differences in house price dynamics across euro area countries (Figure 5). It is striking that, in Germany and Austria, the 5-year moving average growth rates were below 4% or even negative after 1996. The average yearly price increase of the HMR since acquisition between 2000 (2005) and 2010 is approximately 0% (1%) in the case of Germany. In comparison, over the same time horizon, the average yearly price increase in Belgium is 7% (6%). Figure 6 demonstrates that both the country of residence and the year of acquisition strongly influence the average yearly capital gain households were able to obtain if they sold their house today (i.e. at the time of interview). As we will show, this cross-country variation of accumulated HMR price increases is a major factor explaining levels and differences in wealth holdings across euro area countries.

As stated in the introduction, intergenerational transfers are commonly reported to contribute to the wealth accumulation process. The HFCS asks how many substantial gifts or inheritances had been received and collects detailed information of the two to three most significant transfers. These questions allow estimating the value of received gifts or inheritances at the year the household received it (excluding the HMR). Having no knowledge how intergenerational transfers were invested or consumed, we cannot but assume at this stage zero returns of these assets, which we consider, for the lack of other salient alternatives, as the best and most conservative baseline scenario.



Figure 6: Average price increase of residential property for selected years of acquisition

Source: ECB Statistical Data Warehouse. Data sources are written next to each country. The time series "new and existing dwellings" is used for all countries with the exception of Belgium, Finland, France, Netherlands and Slovakia, for which we use "existing dwellings". Countries for which the underlying property price index is not available for the selected years are excluded.

Figure 7 shows the share of households in each country having previously received any substantial gift or inheritance. Figure 8 further distinguishes between households having received their HMR as gift or inheritance (red bar) and households having received a gift or inheritance other than their HMR (blue bar).³ With exception of very low values for Greece and the Netherlands, 20-40% of households report having previously received a substantial gift or inheritance (excluding the HMR). The euro area average (excluding both Finland and Italy) is 28% and thus higher than the figure reported by Wolff and Gittleman (2011) for U.S. American households (21%). For Greece, Italy, Slovenia and Slovakia about 20% or more of households report having inherited or been given the HMR.

Figure 8 depicts the mean contribution of the HMR and intergenerational transfers (both as HMR and other than HMR) to total net wealth. For the euro area aggregate (excluding Finland and Italy), intergenerational transfers excluding the HMR contribute (with their initial value) 11% to total net wealth. Combined with the HMR contribution of 49%, these two factors contribute 60% to the mean total net wealth of households. The pink bars further allow distinguishing the respective contribution of inherited/received HMR and HMR excluding inheritances/gifts. For example, in Austria the mean contribution of the HMR to total net wealth is 42%; 10% emanate from inherited/received HMR (at initial value) and 32% emanate from the HMR excluding inheritances/gifts.





Source: own calculations based on the HFCS UDB 1.0; data are multiply imputed and weighted. No information for Finland on intergenerational transfers, for France on HMR gifts/inheritances and for Italy on gifts/inheritances other than HMR.

³ For Finland and Italy a corresponding question was not asked. For France data on HMR gift or inheritance does not exist in the HFCS.

Figure 8: Mean contribution of the HMR and intergenerational transfers to total net wealth



Source: own calculations based on the HFCS UDB 1.0; data are multiply imputed and weighted. No information for Finland and Italy on intergenerational transfers.⁴ No information for France on whether the HMR was received as gift or inheritance. Euro area figures are not displayed due to different numbers of countries included in the three categories.

3 Residential property price dynamics

Trying to assess the contribution of house price dynamics and thus capital gains from homeownership for household net wealth across countries, it is key to be able to rely on indicators that appropriately reflect past dynamics. Unfortunately, the development of high quality house price indices is plagued by many difficulties and challenges. For example, collecting mean and medians of house price transactions may suffer from compositional changes of transactions over time or transactions may not be fully representative or cease to be representative for the market in question. A key issue is that dwellings are usually transacted very infrequently and repeat sales methods require at least two transactions of a single property, while hedonic price models require large and detailed data sets to correct for quality differences over time (see for example Hilbers et al., 2008 and Case and Wachter, 2005 for details).

Furthermore, available residential property price indices for euro area countries differ rather substantially in scope and availability (Figure 5). For example, residential property price indices for all euro area countries are publicly available for the years 2007-2008 only, as several countries only recently began to publish corresponding indices (Slovenia in 2007, Cyprus in 2006, Slovakia in 2005). The longest time spans are available for Italy (1965-), Belgium (1973-) and Germany (1974-). An additional complication is that the available macroeconomic indices differ in concept. Residential property price indices

⁴ In Italy, the purchase price of the dwelling is not collected for those that have received the HMR as a gift or inherited it (see questionnaire page 26, questions D07-D08). Nevertheless the information is collected for those that have only partially inherited the HMR (182 households). Thus, it is not possible to report correctly the share of the HMR value related to the initial value of the inherited/ gifted HMR.

available for France, the Netherlands, Belgium, Finland and Slovakia refer to existing dwellings whilst for the remaining majority of euro area countries it refers to new and existing dwellings.

In this paper, we will therefore take a different approach. We construct an index of housing value appreciation calculated from self-assessed HMR values for each euro area country using information available in the Eurosystem Household Finance and Consumption Survey.⁵ A similar approach has previously been used by Bucks and Pence (2008). They use data from the U.S. Survey of Consumer Finances and report that U.S. homeowners report house values reasonably accurately. Further, on the positive side, it both avoids the above-mentioned pitfalls and uses local dynamics, specific to the dwelling and household in question and in the sample. It is known that local developments may differ from country-wide developments. Furthermore, constructing a housing value appreciation index based on self-assessment embodies relevant information of both demand and supply conditions, which together shape the development of property prices. Such an index is based on a comparison of the same property over time and is internally coherent as it uses the same data source as used for the calculation of households' net wealth. Moreover, we can harmonise the time span of the index across countries. Finally, the HFCS indices are constructed out of currently-owned HMRs only.6 This ties in nicely with the paper's objective of explaining current net wealth levels and current net wealth differences between countries. Our index includes all HMRs used to calculate the current net wealth for the population sample we are interested in. On the negative side, self-reported house prices are known for being slightly (usually in the order of <10%) biased upward (e.g. Ihlanfeldt and Martinez-Vazquez, 1986; Goodman and Ittner, 1992; Benítez-Silva et al., 2009).

For each country, we calculate an appreciation index to HMR ownership based on selfassessed HMR values from the HFCS data. For each year, we take the average of the estimated current (self-assessed) value or selling price and divide it by the average (selfassessed) acquisition price over all HMRs (either bought or built) in this particular year. The result is a country-specific time-varying index of the average accumulated nominal Housing Value Appreciation since the acquisition of the HMR (henceforth also referred to as mean HVA index). It represents an index based on (non-realised) capital gains from homeownership. Expressed in mathematical terms, for each country C, for each home-

⁵ Using the same methodology as in the present paper, Mathä, Porpiglia and Ziegelmeyer (2014) exploit house price discontinuities at the national border of Luxembourg and report cross-country and spatial differences in house price increases to be the main contributing factor for observed wealth differences in Luxembourg and its surrounding regions.

⁶ The numerator of the HFCS index is based on self-assessed HMR values. The denominator is the value of the HMR at the time of acquisition, thus it can be argued that it is not entirely self-assessed. Only if the HMR was inherited or received as gift, the denominator is also self-assessed. Moreover, there is a self-assessment component if the owner, friends or relatives self-contributed to the HMR. The initial price of the property is something that people remember although it cannot be excluded that memory recall problems exist. However, as long as the assessment is unbiased, memory recall problems are only expected to increase the variance, but do not affect the value itself.

owner household h in the set H_t of households who bought an HMR in year t, we sum over the self-assessed value P at the time of interview T and divide by the sum of the value P at time of acquisition t.

(1)
$$h \in H_t$$
 $r_{t,C}^{\text{HVA}} = \sum_{h \in H_t} P_{T,C}^{\text{HFCS}} / \sum_{h \in H_t} P_{t,C}^{\text{HFCS}}$

Due to the unavailability of the initial value of the HMR in Finland and France housing value appreciation based on HFCS cannot be calculated.⁷ For Slovenia the HVA index is constructed but not shown since the sample size is too small to be considered country representative.

Next, we apply a kernel-weighted local polynomial smoothing (Appendix C, Figure 10). The smoothing is carried out to reduce the influence of outliers as a result of a low number of observations for one or more specific years (Appendix C, Table 8). Figure 9 depicts the development of the smoothed mean HVA index over time for the remaining 12 countries. For reasons of robustness, we additionally construct two more HFCS based HVA indices; the corresponding median index (median HVA index) and a median index calculated as the median over individual accumulated nominal housing value appreciations (median ratio HVA index). The latter is derived for each household as the accumulated nominal HVA from the ratio of the self-assessed value P at the time of interview T to the value P at time of acquisition. Graphically comparing the three HFCS based indices shows their close correspondence (Figure 9).⁸

Figure 9 also depicts the development of the corresponding macroeconomic residential property price index. For this purpose, the index is scaled to 1 in the year of data collection to ensure that indices refer to the same reference year.⁹ On the one hand, we need to take into account that the smoothing of the HVA indices is necessary due to measurement error. On the other hand, smoothing decreases the prevalence of business cycle effects. For countries, such as Belgium, Spain, Greece, Luxembourg, the Netherlands and Portugal the

⁷ The index for Italy only includes the HMR for households who built or purchased the HMR. The purchase price of the dwelling is not collected for those that have received the *complete* HMR as a gift or inherited it. The initial value of the HMR is only available for households who *partially* inherited the HMR (182 households). We drop these 182 households from the index construction for IT due to unknown ownership shares.

⁸ Figure 11 in Appendix C adds to Figure 9 two additional indices: the mean and median HVA indices adjusted by the square metre size of the HMR. The differences compared to the unadjusted mean and median HVA indices are minor and often not even visible in the Figures. Since the square metre size of the HMR is not available in Malta and Portugal, the following analysis focuses on the unadjusted HVA indices.

⁹ In cases where the field phase covered more than one year, the end year is taken. For Greece, we set the index value for the year 2008 to one since the macroeconomic property price index (new and existing dwellings) for the survey year of 2009 is not available yet (latest check 14.04.2014). For most recent years only an index covering new and existing flats is available, which does not cover houses as our other macroeconomic property price indices.

HVA indices are more or less in line with their corresponding macroeconomic index. For Italy, Malta and especially Austria discrepancies are larger.



Figure 9: Housing value appreciation and macro residential property price indices

Source: own calculations based on the HFCS UDB 1.0; data are multiply imputed and weighted. Slovenia is excluded since it has no country representative survey. Information on the initial value of the HMR does not exist for Finland and France. The kernel-weighted local polynomial regression uses the Epanechnikov kernel-function. The kernel bandwidth is 5 and the degree of the polynomial smooth is 1.

An interesting question in this respect is whether differences observed between the HVA index and the macroeconomic index, i.e. the respective over- or under-evaluation of the HMR relative to the macroeconomic index, is related to household and country characteristics. However, this is out of scope of the present paper and we leave this for future research. Furthermore, we set the HFCS based HVA indices to missing if the year of acquisition was prior 1970. Reasons for this are: first, it seems appropriate to exclude all HMRs with acquisition during or before World War II; second, the number of dwellings underlying each annual data point become relatively small prior to 1970 - at least for some countries; third, if the recall bias concerning the initial value of the HMR increases with the time elapsed since acquisition, the most critical years should be excluded; and fourth, we ensure that the time overlap between the macroeconomic and HFCS based indices becomes larger in relative terms. Using identical time periods for the HFCS based HVA and the macroeconomic indices, all key results presented remain unchanged. We use the mean HVA index for the presentation of the results. Results are however generally robust to using other HVA indices or the macroeconomic index. The robustness is explored in more detail in the penultimate section.

The number of household observations using the macroeconomic and HVA indices is shown in Table 1. The reduction of observations for the euro area is similar at about 40%. However, the reduction of observations is much more unevenly distributed for the macroeconomic indices ranging from 7% for the Netherlands to 79% for Slovenia, whereas for the HVA indices, it ranges from 2.5% for the Netherlands to 12% in Spain. The higher overall missing rate of the HVA indices is caused by the exclusion of France since the construction of the HVA indices is not possible (the initial value of the HMR is unknown). Since France is the country with the largest sample size, this strongly affects the number of observations lost. The available time spans of the macroeconomic indices are normally shorter than for the HVA indices. However, the problem is less severe since most house-holds acquired their HMR in recent decades.

				Mac	ro index				HV	'A index		
		Non	Owners w	/o index	Loss of hh	Index	period	Owners w	/o index	Loss of hh	Index	period
Country	Total	owners	with	without	in %	min	max	with	without	in %	min	max
Austria	2,380	1,199	652	529	22.2	1987	2011	1,038	143	6.0	1970	2011
Belgium	2,327	602	1,437	287	12.4	1973	2010	1,501	224	9.6	1970	2010
Cyprus	1,237	247	238	752	60.8	2006	2010	943	47	3.8	1970	2010
Germany	3,565	1,552	1,675	338	9.5	1975	2011	1,828	185	5.2	1970	2011
Spain	6,197	809	3,562	1,826	29.5	1980	2009	4,641	747	12.0	1970	2009
Finland				no year o	f acquisition	availabl	e		no year o	f acquisition	available	9
France	15,006	5,003	7,560	2,443	16.3	1980	2010		no initial v	alue of HMF	R availab	le
Greece	2,971	985	796	1,190	40.1	1994	2008	1,699	287	9.7	1970	2009
Italy	7,951	2,315	5,070	566	7.1	1965	2010	4,763	873	11.0	1970	2010
Luxembourg	950	285	584	81	8.5	1974	2010	618	47	4.9	1970	2011
Malta	843	200	481	162	19.2	1980	2010	593	50	5.9	1970	2010
Netherlands	1,301	337	878	86	6.6	1976	2010	932	32	2.5	1970	2010
Portugal	4,404	1,349	1,644	1,411	32.0	1988	2010	2,732	323	7.3	1970	2010
Slovenia	343	58	15	270	78.7	2007	2010	255	30	8.7	1970	2010
Slovakia	2,057	466	386	1,205	58.6	2005	2010	1,530	61	3.0	1970	2010
Euro area	51,532	15,408	24,977	19,721	38.3	1,965	2,011	23,074	21,625	42.0	1970	2011

Table 1: Availability of housing value appreciation and residential property price indices

Source: own calculations based on the HFCS UDB 1.0; data are multiply imputed and weighted.

4 Estimation technique

First, we assess whether homeownership, house price developments (using either the HVA or macroeconomic residential property price indices) and the amount of intergenerational transfers received in the past influence the median level of total net wealth. We run separate regressions for each country to explore whether these factors individually matter in each country. Second, we use decomposition techniques described in Fortin, Lemieux and Firpo (2011) to analyse the respective contribution of the three factors in explaining cross-country differences in household wealth in the euro area.

The intuition behind the decomposition analysis is that the difference between two groups in a relevant statistic (e.g. mean, median, 75th and 90th percentile) of the variable of interest (in our context the household net wealth) can be broken down into differences in the level of a set of covariates and into differences in the size of the coefficients on the aforementioned set of covariates. In the literature and in the remainder of the paper, the contribution of the former is referred to as *"explained part"*, which means the part of the net wealth difference explained by observable differences in the two populations, while the latter is referred to as *"unexplained part"*, which means the part of the net wealth difference that is not explainable by observable characteristics. The repartition of the household net wealth differences in these two components is done via counterfactual analysis.

At mean level we make use of the Oaxaca-Blinder (Oaxaca, 1973; Blinder, 1973) decomposition. The right skewness of the household total net wealth distribution as well as the varying importance of the HMR and inheritance in the household portfolio along the net wealth distribution require undertaking the counterfactual analysis not only at mean level but also at various points of the net wealth distribution. In the context of our analysis this objective is accomplished using the Oaxaca-Blinder decomposition applied to the recentred influence function of the median, the 75th and the 90th percentile (Fortin, Lemieux and Firpo, 2011; Firpo, Fortin and Lemieux, 2007, 2009) (thereafter RIF-OB decomposition). Other distribution decomposition techniques, which have been proposed in literature (see among others Juhn, Murphy and Pierce, 1993; Di Nardo, Fortin and Lemieux, 1996; Machado and Mata, 2005), can either not or only partially break down the endowment and coefficient effects over household net wealth differences for the vector of covariates, whereas the RIF-OB decomposition is able to provide a detailed decomposition of the wealth gap along the whole household net wealth distribution fully comparable with the Oaxaca-Blinder decomposition.

Coefficients and standard errors presented hereafter are adjusted to account for the multiply imputed nature of the database following Rubin's (1987) combination rules. Results are weighted to take into account the complex survey design structure if not indicated otherwise. We excluded Slovenia from the analysis due to the small sample size and being non-representative. Since information on the initial value of the HMR is missing for France we have to solely rely on the macroeconomic index.

4.1 Median regression

We firstly address whether homeownership and housing value appreciation are able to explain the level in total net wealth in each of the euro area countries. We also provide some pooled regressions for the whole sample. In addition, we are interested in the contribution of received gifts and inheritances on total net wealth controlling for covariates commonly used in wealth regressions (see, among others, Gale and Pence, 2006; Sinning, 2007; Bauer et al., 2011). Secondly we assess whether there are substantial differences between euro area countries with respect to the contribution of these factors on total net wealth. For this purpose we estimate a median regression of the following form:

(2) $W = \beta_0 + \beta_1 Z + \beta_2 E + \beta_3 Y + \beta_4 I + \beta_5 T + \varepsilon,$

where, omitting the household identifier i, W represents total net wealth of each household in the sample, Z stands for a set of household or reference person¹⁰ characteristics such as gender, age and age squared, household size, civil status (single, couple, divorced or widowed) and a immigration dummy indicating whether or not the reference person is born outside the country of residence.¹¹ E is a vector of dummies representing the education level (low, middle, high). Y is a vector representing total household income, employment status (employee, self-employed, unemployed, retired and other), a dummy for having a temporary contract and two dummies for working in the financial or public sector. I contains a homeownership dummy and the mean Housing Value Appreciation (HVA) index introduced in the previous section. The indices used in the multivariate analysis are constructed for both the HFCS and the macroeconomic data in the following way: The variable takes the value zero for non-homeowners and takes the value of the HVA index corresponding to the year of the HMR acquisition. The intuition behind this variable is that homeowners profit from appreciation in the value of their HMR over time. The increase in the housing value represents the capital gains of the investment into homeownership if the household were to sell the HMR at the time of the interview. HMR owners that acquired their HMR earlier, thus, are expected to receive higher capital gains, ceteris paribus. Finally, T includes the received amount of gifts and inheritances (including the HMR)¹² and a variable indicating the time passed in terms of number of years since the largest gift or inheritance was received. $^{\rm 13}$ ϵ is the error term which is assumed to be i.i.d. $^{\rm 14}$

¹⁰ The reference person of each household refers to the "financially knowledgeable person" (FKP), i.e. the person who knows best about the finances of the household.

¹¹ The country of birth is unavailable for Spain, France and the Netherlands. For those countries where this variable is available, it is included in the country specific analysis. In the decomposition analysis, it is excluded for the comparisons of Spain, France and the Netherlands with Germany.

¹² For France, information on whether the HMR was inherited or gifted is unavailable.

¹³ Due to the arising endogeneity issue, we decided not include dummies for various total net wealth quintiles to proxy for different household behaviour or attitudes along the wealth distribution.

¹⁴ The model presented in the paper controls for returns on inherited wealth by including the variable "number of years since the largest gift or inheritance was received". This specification requires the weakest assumptions. The presented results are robust to assuming zero real return on inheritances (i.e. returns on inherited wealth equal inflation) instead. These results are available from the authors upon request.

No. obs. male		,	I C		The	OTECCE	fild	Luxemp.	Malta	Netherlands	r ortugal	Slovenia	SIOVAKIA
male	2218	1956	974	3380	5436	2613	7071	903	785	1243	3933	295	1996
	43.9%	53.5%	57.5%	51.0%	51.3%	40.2%	55.9%	61.0%	53.8%	63.1%	71.2%	41.6%	44.9%
age	50	50	50	51	51	48	54	48	53	51	53	50	47
age2	2756	2778	2774	2909	2817	2533	3175	2513	3029	2862	3091	2711	2433
single	25.8%	20.6%	12.9%	25.5%	13.6%	19.1%	14.0%	26.1%	11.8%	37.0%	11.3%	16.4%	19.8%
divorced	15.3%	14.4%	9.2%	12.6%	8.1%	6.6%	8.4%	14.2%	8.3%	12.8%	8.9%	7.7%	10.0%
widowed	9.4%	10.6%	10.6%	12.0%	12.0%	9.1%	13.3%	6.6%	9.5%	7.4%	12.9%	14.8%	12.9%
hhsize	2.1	2.4	2.8	2.1	2.8	2.7	2.6	2.5	2.9	2.2	2.8	2.6	2.9
mideduc	69.0%	38.0%	31.9%	56.4%	19.8%	39.5%	37.2%	37.9%	22.5%	38.4%	13.8%	54.8%	77.6%
higheduc	14.3%	38.5%	42.0%	29.3%	27.6%	21.8%	11.9%	27.7%	15.3%	33.9%	9.4%	22.6%	16.8%
born in country of residence	88.9%	89.1%	86.4%	85.5%		91.5%	90.5%	54.9%	94.6%		92.8%	90.7%	97.9%
ihs(total income)	10.6	10.4	10.5	10.5	10.3	10.1	10.3	11.2	10.2	10.7	9.8	9.3	9.6
self-employed	10.1%	5.4%	10.3%	6.9%	9.0%	16.7%	11.7%	6.2%	8.1%	3.9%	11.2%	3.6%	7.6%
unemployed	5.1%	10.9%	5.6%	5.4%	10.2%	4.5%	3.8%	2.7%	2.2%	1.7%	8.3%	10.6%	5.6%
retired	32.8%	27.5%	22.1%	27.3%	17.6%	22.9%	34.1%	21.3%	26.0%	20.1%	31.6%	36.1%	22.8%
other	7.1%	9.7%	6.1%	9.7%	19.1%	21.1%	%0 .6	9.6%	25.7%	13.4%	4.4%	6.9%	5.0%
employment status missing	0.0%	2.0%	1.0%						0.1%	13.0%	0.1%		0.5%
temporary contract	2.2%	4.2%	3.5%	6.3%	8.3%	7.9%	4.8%	4.2%	1.6%	4.2%	5.5%	6.5%	5.3%
financial sector	2.2%	2.3%	4.4%	2.7%	2.4%	0.8%	1.7%	10.0%	3.7%	2.6%	1.1%	2.2%	1.8%
public sector	13.0%	17.9%	19.3%	13.0%	12.2%	6.8%	11.3%	18.2%	12.1%	14.7%	10.9%	8.3%	15.4%
owner	44.7%	66.5%	73.3%	41.7%	80.9%	69.1%	65.4%	64.6%	76.4%	56.0%	68.7%	80.3%	89.2%
mean HVA index	0.8	1.9	2.3	0.6	5.0	2.1	2.7	1.8	4.0	1.2	2.3	0.9	2.9
median HVA index	0.8	2.1	2.3	0.6	6.2	2.9	3.2	1.8	4.8	1.2	3.5	1.6	3.5
median ratio HVA index	0.7	2.0	2.3	0.6	6.4	2.6	3.1	1.8	4.8	1.2	3.4	1.8	3.3
ihs(amount gifts & transfers)	3.9	3.4	3.6	3.6	3.0	2.9		3.2	3.2	0.7	2.4	3.9	3.8
years since largest transfer	6.3	3.7	5.1	4.3	3.9	5.3		3.7	3.4	0.6	4.0	6.1	4.6
ihs(net wealth)	9.6	11.0	11.2	9.2	11.0	10.3	11.1	11.4	11.8	8.9	10.2	10.7	10.7

To ease interpretation of the presented results, Table 2 provides some basic household statistics across countries. The variables selected correspond to the covariates included in the regressions and decompositions. The sample is based on the available observations for the mean HVA index.

All monetary units (total net wealth, total income and amount of intergenerational transfers) are transformed using an inverse hyperbolic sine (IHS) transformation in log form (Pence, 2006). This is necessary to avoid convergence problems of the pooled median regression resulting from the large dispersion of total net wealth in combination with large sample sizes.

There has been a long debate on whether or not to use weighted regressions. Due to the complex survey design of the HFCS, weighting is preferable as weighting includes relevant information, such as geographic and operational variables that influence non-response rates across countries. Since geographic and operational variables are not included in the HFCN (at least not to the extent necessary to construct the weights), Faiella (2010) and Magee, Robb and Burbidge (1998) recommend to run weighted regressions. As Stata13 is unable to calculate weighted median regressions, we include the final sampling weights as additional covariate to reduce any potential selection bias normally corrected for by weighted regressions.¹⁵ We address problems related to heteroskedasticity and sampling uncertainty via a new option in Stata13 to calculate robust standard errors for median regressions. We estimate equation (1) for each country separately and in addition in pooled form across countries. In the latter case, we include country fixed effects, where Germany serves as the base country.¹⁶

Table 3 presents the country results of equation (2) where the HMR capital gains are represented by the mean HVA index and the full set of covariates. We briefly note that coefficient estimates of the household specific covariates are as expected. Total net wealth is humped shaped over age for most countries. Being a single, divorced or widowed has usually a negative impact on median total net wealth. The median total net wealth increases with higher education and income. Having a temporary contract and being unemployed normally decreases, while being self-employed increases median net wealth. The results are very much in line with results reported in the household wealth literature. Concerning the variables of main interest, the homeownership dummy has the strongest impact of all explanatory variables and is highly significant for all countries. For example, homeownership increases the expected median total net wealth between 119% (Germany) to 278% (Spain).

¹⁵ We do not report the coefficients and standard errors of the final sampling weight variable in the tables below since we are not interested in the coefficients themselves. In some countries the included final sampling weights were highly collinear with other variables included in the specification. In cases where this leads to the problem of parameter instability, the weight variable was dropped.

¹⁶ We do not run any median regression for Slovenia since the sample size is too small for a country representative sample.

Table 3: Median regression

							Mean									
		(1) AT	(2) BE	(3) CY	(4) DE	(5) ES	(6) FR	(7) GR	(8) IT	(9) LU	(10) MT	(11) NL	(12) PT	(13) SK	(14) EA ex FR	(15) EA ex IT
	male	0.095 ** (0.048)	0.067	0.185 *	0.013	0.107 ***	0.053 ***	0.111 ***	0.067 ***	0.125 **	-0.187 ** (0.073)	0.138	0.254 ***	0.059 *	0.090 ***	0.107 ***
	age	(0.048) 0.047 *** (0.014)		(0.106) 0.080 *** (0.026)		(0.028) 0.041 *** (0.006)			(0.020) 0.028 *** (0.005)		0.024	(0.090) 0.073 *** (0.027)	(0.053) 0.052 *** (0.010)		0.035 *** (0.003)	(0.020) 0.028 *** (0.004)
s	age2	-0.000 ***	-0.000 ***	-0.001 **	-0.000	-0.000 ***	-0.000 ***	-0.000 ***	-0.000 ***	-0.000	-0.000	-0.000 *	-0.000 ***	-0.000	-0.000 ***	-0.000 ***
riable	single	-0.060	-0.139 * (0.081)	-0.290	0.006	-0.056		-0.166 *** (0.055)	-0.072 *	-0.173 * (0.102)	-0.175 (0.140)	-0.253 ** (0.121)	-0.180 ** (0.079)	-0.060	-0.069 *** (0.026)	-0.086 *** (0.031)
hic va	divorced	-0.320 *** (0.121)		-0.333 ** (0.137)	-0.318 *** (0.074)	-0.124 * (0.069)	-0.226 *** (0.035)		-0.179 *** (0.045)		-0.419 ** (0.178)	-0.204 *	-0.232 *** (0.063)		-0.195 *** (0.028)	-0.196 *** (0.034)
ograp	widowed	-0.235 ** (0.095)	-0.248 *** (0.077)	-0.482 * (0.266)	-0.113	-0.010	0.009	-0.159 ** (0.072)	-0.166 *** (0.033)		-0.074 (0.203)	-0.030	-0.154 * (0.090)	(0.002)	-0.144 *** (0.028)	-0.120 *** (0.035)
Socio-demographic variables	hhsize	0.044	0.028	0.008	-0.012 (0.019)	-0.030 ** (0.014)	-0.055 *** (0.009)	0.032 *	-0.023 ** (0.011)	0.042 *	-0.004 (0.036)	0.023	-0.008	0.056 ***	-0.002	(0.033) 0.009 (0.009)
Soci	mideduc	(0.030) 0.405 *** (0.090)		0.332 ** (0.130)	(0.015) 0.386 *** (0.126)	(0.014) 0.296 *** (0.037)	0.304 *** (0.021)	(0.015) 0.347 *** (0.040)	(0.011) 0.297 *** (0.022)	(0.023) 0.111 (0.104)	0.262 *** (0.080)	0.037	(0.022) 0.424 *** (0.044)	0.299 ***	0.350 ***	(0.005) 0.344 *** (0.025)
	higheduc	(0.050) 0.579 *** (0.126)		0.465 *** (0.136)	0.566 ***		0.472 ***		0.505 ***	0.362 *** (0.112)	0.302 ***	0.258 ***	0.879 *** (0.069)	0.580 ***	0.603 ***	0.567 ***
	born in country of residence	0.338 *** (0.126)	0.459 *** (0.076)	0.151 (0.112)	0.300 *** (0.076)	(0.054)	(0.020)	0.552 *** (0.120)	0.625 ***	0.110	(0.002)	(0.075)	0.060	0.112	(0.022)	(0.020)
	ihs(total income)	0.594 *** (0.114)	0.174 ***	0.354 ***	0.796 ***	0.507 ***	0.532 ***	0.354 ***	0.407 ***	0.369 ***	0.385 ***	0.036	0.263 ***	0.305 ***	0.397 ***	0.395 *** (0.014)
	self-employed	(0.114) 0.569 *** (0.117)			(0.039) 0.337 *** (0.073)		(0.018) 0.680 *** (0.031)				(0.062) 0.667 *** (0.147)	(0.055) 0.397 * (0.205)	(0.025) 0.673 *** (0.046)		0.583 *** (0.028)	(0.014) 0.554 *** (0.032)
bles	unemployed	-0.539 * (0.315)	-0.417 *** (0.094)	-0.033	-2.001 *** (0.386)	-0.070	-0.516 *** (0.075)	-0.093	0.184 **	-0.436	0.064	0.056	-0.160 *	-0.202 ** (0.102)	-0.212 *** (0.036)	-0.275 *** (0.042)
l varia	retired	0.063	-0.144 * (0.075)	-0.036	0.177 **	0.094 *	0.007	0.138 **	0.187 ***	-0.012 (0.134)	0.003	-0.219 ** (0.088)	0.120 **	-0.036	0.087 ***	0.044
related	other	0.218	-0.236 ** (0.100)	-0.335 * (0.195)	0.161	0.215 ***	-0.126 **	0.168 ***	0.202 ***	-0.056	-0.140	-0.190	0.149	0.045	0.179 *** (0.029)	0.140 *** (0.034)
Employment related variables	employment status missing		0.089 (0.590)	0.236 (0.544)								0.052 (0.116)	0.397 (0.572)		0.018 (0.124)	0.033 (0.119)
Emplc	temporary contract	0.094 (0.421)	-0.055 (0.194)	0.055 (0.298)	-0.225 (0.226)	-0.182 *** (0.045)	-0.063 (0.065)	-0.099 ** (0.045)	-0.120 ** (0.058)	-0.119 (0.414)	-0.043 (0.477)	-0.047 (0.357)	-0.117 (0.125)	-0.254 *** (0.082)	-0.193 *** (0.037)	-0.185 *** (0.046)
	financial sector	0.032 (0.117)	0.049 (0.125)	-0.337 ** (0.132)		0.108 (0.129)	0.179 *** (0.045)	0.158 (0.152)	0.108 *** (0.038)	0.135 (0.103)	0.103 (0.251)	-0.170 (0.290)	0.121 (0.164)	0.248 *** (0.065)	0.092 *	0.101 *
	public sector	0.082 (0.094)	-0.034 (0.053)	-0.321 *** (0.115)		-0.053 (0.037)	-0.099 *** (0.027)	0.069 (0.048)	0.042	-0.111 (0.088)	0.036	-0.096 (0.097)	0.042	0.045	0.001	-0.013 (0.030)
st	owner	1.875 *** (0.184)	2.426 *** (0.107)	2.537 *** (0.283)	1.189 *** (0.129)	2.781 *** (0.130)	2.265 *** (0.042)		2.487 *** (0.037)	1.956 *** (0.177)	2.087 *** (0.170)	1.896 *** (0.191)	2.724 *** (0.094)	2.446 *** (0.106)	2.543 *** (0.019)	2.415 *** (0.024)
main interest	mean HVA index	0.133 *	0.055 ***	0.051 **	0.438 *** (0.074)	0.004 **		0.070 ***	0.027 ***	0.094 ***	0.019	0.043 *	0.035 ***	0.056 ***	0.017 *** (0.002)	0.016 *** (0.003)
f main	ihs(gifts & transfers)	0.054 ***	0.029 ***	0.058 ***	0.059 ***	0.039 ***	0.053 ***	0.033 ***		0.030 ***	0.036 ***	0.033 **	0.043 ***	0.030 ***		0.042 ***
Variables of	years since largest transfer	-0.006 * (0.003)	0.002 (0.003)	-0.007 (0.005)	-0.010 *** (0.002)	-0.001	-0.000	-0.017 *** (0.002)		-0.005	-0.006 (0.007)	-0.004 (0.011)	-0.008 *** (0.003)	-0.011 *** (0.003)		-0.005 *** (0.001)
Variá	macro index						0.037 ***									
	constant	1.407 (1.271)	5.543 *** (0.495)	3.840 *** (0.756)	0.233 (0.507)	2.740 *** (0.310)	3.340 *** (0.222)	3.838 *** (0.495)	4.074 *** (0.230)	4.896 *** (0.773)	5.422 *** (0.800)	7.108 *** (0.941)	4.263 *** (0.370)	4.621 *** (0.436)	4.200 *** (0.134)	4.249 *** (0.209)
	min. no. of obs.	2213	1954	972	3379	5433	12197	2612	7071	903	785	1236	3929	1996	33231	25723

Mean HVA index

 $\frac{1}{1} \frac{1}{1} \frac{1}$

Acquiring the HMR should not change the household wealth position at the point of transaction since, most likely, financial wealth is exchanged with real estate wealth and mortgage debt. The wealth effect of homeownership is likely to be linked to homeowners having a different consumption and saving behaviour. Before the acquisition of the HMR

they are likely to consume less and save more in order to make required down payments or decrease the loan to value ratio, which generally helps to reduce the interest rate on the mortgage. In addition, they are likely to save more regularly in form of their (often monthly) mortgage repayment. This also is confirmed by results from the German HFCS, which contains information on yearly net saving (Deutsche Bundesbank, 2013, p. 48). The mean (median) saving rate of tenants is 8% (3%). The corresponding saving rate of HMR owners with mortgage is 22% (21%) and without mortgage 13% (7%). The higher saving rate of owners ignores the fact that owners are more likely to have higher income on average and hence a higher possibility of saving. The multivariate analysis takes this into account since the equation includes total income as control. Moreover, they do pay their rent in terms of either interest and/or redemption payments. Finally, interest rate payments on HMR mortgages are tax deductible in most euro area countries.¹⁷

Despite controlling for homeownership, the property price dynamics contribute positively for most countries. Only for Malta, the HVA index seems to play no role. For example, 2010 HMR values in Germany are on average 21% higher than in 2000. The estimated coefficient of the mean HVA index for Germany is 0.438 (see Table 3). This means that a household acquiring the HMR in 2000 has a median net wealth, which is 9.2% higher holding homeownership, income and other controls constant. Italian house prices in 2010 using the mean HVA index are 230% larger than in 1990. The estimated coefficient for Italy is 0.027, meaning that an Italian household acquiring the HMR in 1990 has 6.2% higher median net wealth than a comparable household acquiring the HMR in the year the survey was conducted. These two examples show that the estimated coefficients of the mean HVA index are not only statistically but also economically important.

Gifts and inheritances significantly increase total net wealth in all countries. A ten percent increase in the inherited amount increases median net wealth between 0.29% (Belgium) and 0.59% (Germany). The additional control variable "time elapsed since the largest transfer was received" is insignificant for six and for the other six countries it has a negative impact. On the one hand, this negative effect might be a surprise since households had more time to earn returns on the inheritance. We do not know what households did with the additional wealth; they also had more time to consume the inheritance in total or in part, for example by buying a car. Judging on the basis of the estimated coefficients, it seems indeed that the latter effect dominates. The results carry over to the pooled median regression for the euro area (spec. 14 and 15). The effects of the control variables are as described before. Homeownership increases median net wealth by between 242% and 254%.

To sum up, these results provide compelling evidence that homeownership, capital gains to property and received inheritances matter a great deal for the wealth accumulation process in the euro area and in most euro area countries individually.

¹⁷ The exceptions are Germany, Cyprus, Malta and Slovenia (see Table 3 in ECB (2009), p. 35).

4.2 Oaxaca-Blinder decomposition

In the second part of the analysis, we apply the widely used Oaxaca-Blinder decomposition (Oaxaca 1973, Blinder 1973) to quantify how much of the observed mean household total net wealth differences are due to the difference in covariates – *the explained part* – and how much is due to the differences in the covariates' coefficients – *the unexplained part* (Jann, 2008). For each country, we compute the household total net wealth difference with respect to Germany. Germany is a natural choice as reference country since on the one hand it is the largest country in the euro area with respect to population and total GDP and on the other hand it has the lowest median net household wealth in the sampled euro area countries. In addition, German data includes all the variables included in the model. All these facts ease the economic interpretation of the results given the decomposition coefficients (for what concerns the household net wealth differences) have the same sign for all the considered countries and no variables need to be dropped from the model because they are missing in the country of reference.

The size of the unexplained part in the Oaxaca-Blinder decomposition can differ across euro area countries because of differences in institutional aspects (e.g. differences in fiscal measures such as subsidies and taxes, financial market development and deregulation, banking supervision, pensions and social security), but also aspects related to differences in housing markets (e.g. provision of social housing, the regulation of the rental market, transaction cost differences, competition between mortgage banks).

Relying on the notation as outlined by Jann (2008) the two-fold Oaxaca-Blinder decomposition of the mean household total net wealth gap between a euro area country (CT) and Germany (DE) can be written as:

(3)
$$R = \underbrace{(\overline{X}_{CT} - \overline{X}_{DE})'\widehat{\beta}_{CT}}_{\text{Endowment effect/explained part}} + \underbrace{\overline{X}'_{DE}(\widehat{\beta}_{CT} - \widehat{\beta}_{DE})}_{\text{Coefficient effect/unexplained part}},$$

where *R* represents the household total net wealth difference between the country in question and Germany, which implies in our context that wealth differences are usually positive. *X* is the set of relevant covariates as employed in the median regression and β is the relative vector of coefficients; the subscript *CT* and *DE* indicate respectively the country of interest and Germany. Furthermore, relying on the algebraic properties of the Oaxaca-Blinder decomposition, it is possible to identify the contribution of each endowment difference and each coefficient difference to the household total net wealth difference.¹⁸

¹⁸ As indicated in the first term of equation (3), we use the coefficients of the country (CT), which we compare to Germany, to calculate the endowment effect. This complies with the best practice in decomposition analysis since the range of realisations of the dependent variable in CT includes the range of realisations in Germany.

The Oaxaca-Blinder decomposition assumes, as all the OLS based methodologies, the quasi-normality of the dependent variable (e.g. Cobb-Clark and Hildebrand, 2006; Gale and Pence, 2006). Due to the substantial skewness of the net wealth distribution and to reduce possible biases introduced by outliers, we transform the dependent variable and all the monetary explanatory variables (total income and the amount of intergenerational transfers) in log form, using an inverse hyperbolic sine transformation. A further advantage of this non-linear transformation is that the assumed linearity in the Oaxaca-Blinder decomposition between the dependent and the independent variables is relaxed (Barsky et al., 2002).

For the clarity of exposition, Table 4 shows covariates grouped into various categories: *"demographic"* includes household characteristics such as gender, age, age squared, marital status, household size, educational attainment and where available the variable *"born in the country of residence"*; *"employment"* includes dummies of the employment status, a dummy for having a temporary contract and two dummies for working in the financial or public sector. All other variables are separately reported since they are in the key interest of this study. We address the possible concerns related to heteroskedasticity and sampling uncertainty calculating bootstrapped standard errors over 500 replicate weights (Cameron and Trivedi, 2010).

The decomposition results presented in Table 4 show that (IHS transformed) mean household total net wealth is significantly higher in all countries (with the exception of the Netherlands) than in Germany. Differences in endowments play a major role in explaining wealth differences between Germany and respectively Belgium, Cyprus, Spain, Greece and Slovakia. For the remaining countries, differences in endowments' returns are more relevant.

For all euro area countries, differences in homeownership rates help to explain wealth differences to Germany. Differences in house price dynamics (as captured by the mean HVA index) positively contribute in explaining wealth differences between Germany and Cyprus, Spain, France, Greece, Italy, Portugal and Slovakia, respectively. The higher gross income of German households reduces the net wealth differences with respect to Greece, Italy, Malta, Portugal and Slovakia. If German households had the same gross income as the households in these countries, the net wealth difference would have been even larger than currently observed. Intergenerational transfers and the years since the household obtained the largest transfer play no role in explaining net wealth differences. In Spain, Greece and Portugal they even contribute negatively.

Turning to the effect of endowments' returns on household wealth differences, the interpretation of the values reported for homeownership and the HVA index are not straightforward. The first represents the effect of the HMR capital gain not reflected by the endowment effect of the HVA index on the difference of household total net wealth between Germany and the respective country in question. The latter represents the effect of a return over the HMR capital gain, being conceptually similar to a squared term effect. The residual effect of homeownership (after controlling for the HVA index) on the household total net wealth difference between Germany and the respective country is positive and statistically significant in most of the countries, which suggests for those countries that households investing in homeownership might may have enjoyed benefits not related to capital gains in the context of property price appreciations (e.g. subsidies or tax benefits such as deductible interest rates payments for mortgages etc.). The unexplained part relative to the HVA index is negatively significant only once; this being the case for Malta (Table 4).

Thus, the results of the Oaxaca-Blinder decomposition reinforce the results obtained from the median regressions. While in the median regressions homeownership and property price dynamics were important factors helping to explain net wealth levels, the mean decomposition shows that the differences in these two factors also strongly contribute in explaining mean household total net wealth differences between euro area countries (and Germany).

				- IV	lean HV	'A index	; -					
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
	AT	BE	CY	ES	FR	GR	IT	LU	MT	NL	РТ	SK
overall												
country of interest	9.915 ***	10.970 ***	11.240 ***	11.030 ***	10.160 ***	10.340 ***	11.150 ***	11.400 ***	11.770 ***	8.914 ***	10.160 ***	10.730 ***
	0.142	0.125	0.190	0.119	0.065	0.089	0.069	0.177	0.090	0.258	0.090	0.044
Germany	9.188 ***	9.188 ***	9.188 ***	9.188 ***	9.092 ***	9.188 ***	9.188 ***	9.188 ***	9.188 ***	9.188 ***	9.188 ***	9.188 ***
	0.153	0.153	0.153	0.153	0.158	0.153	0.153	0.153	0.153	0.153	0.153	0.153
difference	0.727 ***	1.785 ***	2.057 ***	1.845 ***	1.073 ***	1.147 ***	1.958 ***	2.210 ***	2.583 ***	-0.274	0.972 ***	1.537 ***
	0.210	0.202	0.252	0.193	0.174	0.178	0.171	0.234	0.187	0.311	0.181	0.158
explained	0.023	0.910 ***	1.716 ***	1.763 ***	0.181 ***	1.223 ***	0.644 ***	0.877 ***	0.858 ***	0.295	0.283 **	2.072 ***
	0.120	0.122	0.274	0.180	0.063	0.120	0.064	0.251	0.159	0.337	0.143	0.203
unexplained	0.705 ***	0.875 ***	0.340	0.081	0.891 ***	-0.076	1.314 ***	1.333 ***	1.725 ***	-0.569	0.688 ***	-0.534 **
	0.200	0.230	0.385	0.271	0.171	0.228	0.172	0.347	0.241	0.434	0.211	0.263
explained												
mean HVA index	0.049	0.102	0.329 ***	0.094 *	0.053 *	0.216 ***	0.100 ***	0.130	-0.072	0.164	0.128 ***	0.181 ***
	0.041	0.085	0.108	0.054	0.029	0.062	0.026	0.097	0.094	0.116	0.036	0.049
owner	0.082 **	0.988 ***	1.230 ***	1.874 ***	0.326 ***	1.148 ***	0.767 ***	0.901 ***	1.226 ***	0.325 ***	0.922 ***	2.023 ***
	0.041	0.114	0.219	0.166	0.030	0.100	0.044	0.136	0.146	0.119	0.077	0.143
ihs(total income)	0.033	-0.024	-0.004	-0.018	-0.010	-0.160 ***	-0.075 ***	0.201	-0.194 ***	-0.023	-0.145 ***	-0.225 **
	0.033	0.015	0.012	0.021	0.007	0.029	0.018	0.166	0.051	0.027	0.035	0.088
employment status	0.006	-0.093 **	0.028	0.007	0.002	0.089 **	0.012	0.038	-0.087	-0.153	-0.005	0.004
	0.043	0.045	0.067	0.039	0.018	0.044	0.019	0.080	0.056	0.132	0.025	0.017
demographics	-0.158 **	-0.053	0.157	-0.142	-0.215 ***	-0.018	-0.160 ***	-0.382 **	-0.012	0.256	-0.514 ***	0.084
	0.063	0.064	0.193	0.124	0.043	0.114	0.048	0.192	0.118	0.234	0.111	0.106
ihs(gifts & transfers)	0.036 0.025	-0.003 0.008	-0.001 0.029	-0.055 *** 0.019	0.022 0.014	-0.034 * 0.018		-0.025 0.017	-0.015 0.013	-0.331 0.243	-0.107 *** 0.024	0.008 0.008
years since largest	-0.026	-0.008	-0.023	0.003	0.002	-0.018		0.014	0.011	0.057	0.005	-0.003
transfer	0.020	0.009	0.018	0.003	0.005	0.012		0.016	0.011	0.180	0.005	0.004
unexplained												
mean HVA index	-0.228	-0.312	-0.245	-0.345	-0.259	-0.275	-0.175	-0.298	-0.370 *	-0.195	-0.315	-0.311
	0.247	0.222	0.223	0.224	0.193	0.222	0.215	0.229	0.222	0.258	0.221	0.220
owner	0.300	0.835 **	0.801 **	1.171 ***	0.601 **	0.926 ***	0.188	0.814 **	0.650 *	0.125	0.601 **	0.954 ***
	0.342	0.332	0.367	0.332	0.262	0.329	0.272	0.358	0.337	0.458	0.288	0.307
ihs(total income)	-0.384	-5.252 **	-6.181 **	-6.232 **	-4.984 **	-2.992	-2.521	-3.704	-0.505	-8.453 ***	-4.911 **	-4.474 *
	2.900	2.610	3.090	2.525	2.524	2.580	2.528	3.446	2.869	2.912	2.501	2.592
employment status	0.358	0.263	-0.069	0.443	0.399 *	0.521 **	0.317	0.269	0.311	-0.866 *	0.468 **	0.377 *
	0.283	0.269	0.386	0.277	0.228	0.265	0.204	0.407	0.235	0.456	0.229	0.216
demographics	0.755	1.514	-0.717	2.430	-1.796	-1.201	-0.894	-0.337	-1.153	17.360 ***	0.981	-1.967
	2.263	2.368	2.697	2.512	1.964	2.067	1.915	3.116	2.349	4.295	2.388	2.006
ihs(gifts & transfers)	-0.201 0.132	-0.553 *** 0.137	-0.217 0.137	-0.283 ** 0.119	-0.326 *** 0.119	-0.416 *** 0.127		-0.382 *** 0.127	-0.453 *** 0.113	-0.204 0.320	-0.286 ** 0.117	-0.503 *** 0.110
years since largest transfer	0.095 0.063	0.212 *** 0.069	0.030 0.083	0.121 ** 0.055	0.149 *** 0.054	0.069 0.064		0.060 0.090	0.098 0.067	0.084 0.215	0.092 * 0.053	0.108 ** 0.054
constant	0.009	4.169	6.938 **	2.777	7.108 ***	3.293	4.399	4.912	3.147	-8.416 *	4.060	5.282 *
	3.130	3.017	3.375	3.123	2.706	2.784	2.785	4.179	3.204	5.112	2.972	2.804
minium no. obs.	5592	5333	4353	8814	15423	5993	10450	4282	4164	4617	7310	5375

Table 4: Oaxaca-Blinder decomposition - Mean HVA index -

Source: own calculations based on the HFCS UDB 1.0; results adjusted for multiple imputation and bootstrapped standard errors with 500 replicates; * p < 0.1 ** p < 0.05 *** p < 0.01. All monetary values (total net wealth, total income and intergenerational transfers) are transformed using an inverse hyperbolic sine transformation in log form. For France, the macro index is used. Variables are grouped as follows: Employee status (self-employed, unemployed, retired, other employment status, temporary contract, financial sector, public sector), demographics (male, age, age2, single, divorced, widowed, hhsize, mideduc, higheduc, born in the country of residence). Household total net wealth differences are relative to Germany.

4.3 RIF-OB decomposition

The recentred influence function (RIF) OB decomposition allows obtaining detailed information on the factors explaining differences between our two sub-populations for the entire distribution of the variable of interest. While a detailed explanation of the RIF-OB decomposition is beyond the scope of this paper, it is worth to introduce some general notions. As detailed in Firpo, Fortin, Lemieux (2007, 2009) and in Fortin, Lemieux and Firpo (2011), by replacing the variable of interest (in our case the household total net wealth) with the recentred influence function of a specific quantile, it is possible to link a distributional analysis to a standard regression framework. Given the properties of the recentred influence function (RIF), it is possible to model the expected value of RIF of the percentile of interest as a linear function of a set of covariates.

(4) $E[RIF(W;q) | X] = X\gamma + \varepsilon$

where *W* is the household total net wealth, *q* is the percentile of interest, *X* is a vector of covariates and γ is a vector of coefficients. Equation (4) can be estimated via OLS for the respective subpopulation and therefore it is possible to apply an Oaxaca-Blinder decomposition analogous to the decomposition presented in equation (3). As the Oaxaca-Blinder decomposition, the RIF-OB decomposition can provide a covariate breakdown of the coefficient estimates, it is robust to the inclusion of non-binary covariate sets and it is not sensitive to the ordering of the covariates (Fortin, Lemieux and Firpo, 2011; Firpo, Fortin and Lemieux, 2007, 2009). To increase the comparability with the Oaxaca-Blinder decomposition presented in Table 4, the dependent variable is transformed using an IHS in log form and we use the same set of covariates and the same reference group as before.

Table 5 shows the results of the RIF-OB decomposition at 50th, 75th and 90th quantile, as well as the mean from the OB decomposition from Table 4 for the variables of main interest (homeownership rate, mean HVA index, amount of gifts and transfers and years since the largest transfer). As for the OB decomposition, heteroskedasticity and sampling uncertainty concerns are addressed, calculating bootstrapped standard errors over 500 replicate weights for each of the proposed specifications (Cameron and Trivedi, 2010).

At *median* level the difference in (IHS) total net wealth levels between the country in question and Germany are always statistically significant a 99% level (95% for Austria). For Austria, France, the Netherlands and Slovakia, wealth differences are mainly explained by differences in endowments (note that the unexplained part is the difference between the total difference and the explained part). For the remainder of the countries, differences in endowments, while not being most important, still contribute substantially to the explanation of differences in IHS net wealth (with the exception of Portugal).

Juai
and 90 th c
÷
. 50 th 75
nposition at mean, 5
ion at
nposit
decon
RIF-OB
und RIF
: OB a
Table 5

					- Mean	Mean HVA index –	-xer		Т				
		(1)	(2)	(3)	(4)	(2)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
		AT	BE	Ç	ES	FR	GR	IT	ΓΩ	МТ	NL	ΡT	SK
	difference	0.73 ***	1.79 ***	2.06 ***	1.85 ***	1.07 ***	1.15 ***	1.96 ***	2.21 ***	2.58 ***	-0.27	0.97 ***	1.54 ***
	explained	0.02	0.91 ***	1.72 ***	1.76 ***	0.18 ***	1.22 ***	0.64 ***	0.88 ***	0.86 ***	0.30	0.28 **	2.07 ***
	mean HVA index	0.05	0.10	0.33 ***	0.09 *	0.05 *	0.22 ***	0.10 ***	0.13	-0.07	0.16	0.13 ***	0.18 ***
duı [-və	owner	0.08 **	*** 66.0	1.23 ***	1.87 ***	0.33 ***	1.15 ***	0.77 ***	0.90 ***	1.23 ***	0.33 ***	0.92 ***	2.02 ***
	ihs(gifts & transfers)	0.04	0.00	0.00	-0.06 ***	0.02	-0.03 *		-0.03	-0.01	-0.33	-0.11 ***	0.01
	years since largest transfer	-0.03	-0.01	-0.02	0.00	0.00	-0.02		0.01	0.01	0.06	0.00	0.00
	difference	0.32 **	1.38 ***	1.56 ***	1.33 ***	0.68 ***	0.82 ***	1.26 ***	2.00 ***	1.48 ***	0.75 ***	0.45 ***	0.30 ***
юц	explained	0.18 **	0.54 ***	0.64 ***	0.22 ***	0.44 ***	0.36 ***	0.37 ***	0.64 ***	0.35 ***	0.41 ***	0.03	0.40 ***
iso	mean HVA index	0.11 ***	0.29 ***	0.16 ***	0.07 ***	0.21 ***	0.07	0.06 ***	0.34 ***	0.17 ***	0.24 ***	0.04	0.21 ***
	owner	0.11 **	0.25 ***	0.30 ***	0.31 ***	0.32 ***	0.35 ***	0.41 ***	0.24 ***	0.30 ***	0.22 ***	0.40 ***	0.23 ***
Ч‡(05ә 8	ihs(gifts & transfers)	0.04	-0.01	0.00	-0.03 ***	0.03	-0.03 **		-0.02	-0.01	-0.11 *	-0.06 ***	0.01
	years since largest transfer	-0.03 *	0.00	-0.01	0.00	-0.01	-0.02 **		0.00	0.01	0.05	0.01	0.00
	difference	0.21 ***	0.69 ***	1.03 ***	0.51 ***	0.31 ***	0.04	0.46 ***	1.23 ***	0.68 ***	0.24 ***	-0.21 ***	-0.64 ***
юij	explained	0.08	0.24 ***	0.31 ***	0.00	0.11 ***	0.12 ***	0.06 *	0.28 ***	0.04	0.11	-0.23 ***	0.10
iso		0.01	0.15 ***	0.07	0.06 ***	0.14 ***	0.05	0.07 ***	0.19 ***	0.11	0.24 ***	0.00	0.17 ***
Pen Jung ZIF-	owner	0.04 **	0.06 **	0.17 ***	0.12 ***	0.05 ***	0.14 ***	0.14 ***	0.03	0.11 **	0.06 ***	0.20 ***	*** 60.0
)9CC	ihs(gifts & transfers)	0.02	-0.01	0.00	-0.03 ***	0.02	-0.02		-0.03	-0.02	-0.21 ***	-0.04 ***	0.01
	years since largest transfer	-0.01	0.00	-0.01	0.00	-0.01 *	-0.01		0.01	0.00	* 60.0	0.00	0.00
		0.22 ***	0.45 ***	1.11 ***	0.35 ***	0.17 ***	-0.19 **	0.27 ***	1.08 ***	0.47 ***	-0.01	-0.38 ***	-0.98 ***
юц	explained	* 60.0	0.07 *	0.14	-0.11 *	0.01	-0.01	-0.01	0.19 *	-0.07	-0.18 **	-0.43 ***	-0.15
aO- isoc rcei	mean HVA index	-0.02	0.01	0.12	0.04	0.06 ***	0.00	0.06 **	0.11	0.09	0.11 ***	0.00	* 60.0
łwo	owner	0.03 *	0.05 *	0.06	0.07 **	0.03 ***	0.09 ***	0.08 ***	0.03	0.02	0.04 **	0.12 ***	0.05
)))))	ihs(gifts & transfers)	0.02	0.00	0.00	-0.04 ***	0.02	-0.01		-0.02	-0.02	-0.28 ***	-0.04 ***	0.01
	years since largest transfer	0.00	0.00	-0.02	0.00	-0.01 *	0.00		0.01	-0.01	0.04	0.00	0.00
Source: 07 p<0.01. A For France	Source: own calculations based on the HFCS U p<0.01. All monetary values (total net vealth, For France, the macro index is used. Variables		IDB 1.0; results adjusted for multiple imputation and bootstrapped standard errors with 500 replicates; * p<0.1 ** p<0.05 *** total income and intergenerational transfers) are transformed using an inverse hyperbolic sine transformation in log form. are grouped as follows: Employee status (self-employed, unemployed, retired, other employment status, temporary contract, and and soon sincle discound discound theirs withdus biologics bow in the construct of variance).	Idjusted for l intergene. Allows: Emp	r multiple i rational tra ployee statu	mputation msfers) are us (self-em	and bootst e transform ployed, un	rapped star ted using a employed, 1	ndard error n inverse h etired, oth	s with 500 uperbolic s er employn	replicates; ine transfo tent status,	* p<0.1 ** prmation in temporary	9<0.05 *** log form. r contract,
wealth diff	winners occurs, particulation to Germany, with the control of the second s	6.1	ווווני, ופרד איי ארב איי אייטי בני אייטי בני אייטטרני אוואבט, אוואבטיר אוואניים איי איי איי איי איי איי איי איי	11810, 4100	icca, wiaco	UCH, 111012C	, 111111CU 10.	III Sucauci		common do	ורסווורור)	. 110490101	4 10141 1101

ECB Working Paper 1690, July 2014

Differences in the homeownership rates explain a substantial part of the wealth difference between Germany and the respective country at median level. In analogy with the Oaxaca-Blinder decomposition, the median RIF-OB decomposition shows that net wealth differences are significant at the 95% level or better in all countries (relative to Germany) with the exception of Portugal. Differences in house price dynamics and thus non-realised capital gains as synthesised by the HVA index have a strong effect on the household median net wealth differences in the analysed countries (with the exception of Greece and Portugal). With respect to the contribution to difference in net wealth, the HVA index is for a minority of countries more important than the HMR ownership indicator variable itself (Belgium, Luxembourg, the Netherlands), in other countries roughly as important (Austria) and for most countries less important than the HMR ownership (Cyprus, Spain, France, Greece, Italy, Malta, Portugal, Slovakia).

In most countries, differences in the amount of gifts and inheritances received by households (relative to German households) seem to be of low relevance for explaining net wealth differences at median level. For Spain, Greece, the Netherlands and Portugal, it contributes negatively, and significantly so, to explaining net wealth differences with Germany. This means that the wealth difference would have been even larger if German households had received the same amount of gifts and inheritances as household populations in these countries.

At the 75th quantile, the difference in (IHS) total household net wealth between each considered country and Germany is generally lower than at median level. The explained part is significant in about one half of the countries (7 out of 12). The homeownership dummy contributes (to the difference in IHS household total net wealth) in 11 out of 12 euro area countries. The average contribution of the four factors (the summed four coefficients divided by the difference in IHS net wealth) across euro area countries is 20% (neglecting those countries where the difference in IHS net wealth is insignificant), and thus smaller than at the median level. This may reflect the decreasing importance of the HMR in the household portfolio at the upper end of the net wealth distribution (see Figure 4). Supporting this argument is that differences in the HVA index contribute significantly to the difference in (IHS) total household net wealth in 7 out of 12 euro area countries, thus three countries less than at the median level. Overall, the significance and contribution of the inheritances to the difference in IHS household total net wealth between each analysed country and Germany stays modest and in line with what was observed at median level.

At the 90th quantile, the explained part is significant in 6 out of 12 euro area countries. For some countries, the difference in IHS wealth is negative as is the explained part, reflecting in part the high skewness of the German net wealth distribution. Furthermore, the explained part is insignificant for Cyprus, France, Greece, Italy, Malta and Slovakia. At this point of the household net wealth distribution, the differences in the factors encompassed in the explained part have difficulty to capture the difference in IHS wealth. Consequently, the unexplained part or the differences in the coefficients tends to become more important.

This suggests that wealth differences in euro area countries in wealthier strata of the population may be influenced by factors other than those relevant at lower strata of net wealth distribution.

In line with the decrease of both the difference in IHS wealth and the overall explained part, the importance of differences in homeownership rates decreases when we move through the household net wealth distribution from bottom to top; similarly differences in property price dynamics become less relevant. Thus, just for a handful of countries both the difference in IHS net wealth and the overall explained part remain significant at the 90th percentile (Austria, Belgium, Spain, the Luxembourg and Portugal); This suggests that homeownership and the HVA index are better in explaining wealth differences at middle part of the net wealth distribution than at the upper tail. At the upper tail, household total net wealth differences in the euro area seem to become more erratic, meaning that we are note able to capture them as well with our factors of interest. The generally lower contributions of differences in homeownership rates and property prices are not much of a surprise, as differences genuinely diminish and HMR related wealth looses importance relative to other real wealth and financial wealth in household portfolios.

In summary, the RIF-OB decomposition supports the findings at mean level. At the respective median, a large part of the observed household total net wealth differences between euro area countries and Germany is attributable to differences in homeownership rates and in property price dynamics. The relevance of differences in property price indices and homeownership rates in explaining net wealth differences tends to decrease toward the upper tail of the net wealth distribution. At the median net wealth level, the aggregate difference in endowments of the four factors of main interest explains on average 56% across euro area countries of the differences in (IHS) household total net wealth (always relative to Germany). This shrinks to 20% at 75th percentile, and to 11% at 90th percentile. This is coincides with the reduced relevance of the endowment effects towards the upper end of the net wealth distribution.

5 Robustness

The main results make use of the HFCS based mean housing value appreciation index. To assess the robustness of the results we run a battery of checks where we essentially replace the mean HVA index with the median or the median ratio HVA index. Overall, the results are very similar, regardless of whether we refer to the median regression, the Oaxaca-Blinder decomposition or the RIF-OB decomposition. Tables referring to these estimations are provided in Appendix D. Furthermore, we replace the mean HVA index with the mac-roeconomic residential property price index. Here, more differences are discernible, which is partially related to the sometimes much smaller sample size at our disposal.

Median regression

At the median regression, the coefficients with regard to homeownership and intergenerational transfers are very robust (Table 9) regardless of the index used. The size of the property price coefficient is different from the baseline specification for most countries using the macroeconomic index. Turning to the pooled estimation results, excluding Italy from the sample to include the inheritance variables increases the coefficient for the macroeconomic property price index from 0.5% to almost 3.3% (Table 10). This is likely to be linked to the fact that the time period for the macroeconomic index is very long in Italy. The values for the macroeconomic property price index in the early years (1965-1967) are above 50. If Italy is excluded, these very influential observations are dropped and the impact of the other observations increases. Using the mean, median, or median ratio HVA indices instead has less of an influence. If the HVA index is increased by 1 unit, median net wealth increases by roughly 1-2%. Despite controlling for homeownership, the property price dynamics contribute to explaining net wealth levels for most countries. Robust evidence of the influence of the house price index over all specifications is obtained for Belgium, Germany, Greece, Italy, Luxembourg, the Netherlands, Portugal and Slovakia.

Decomposition analysis

Turning to our key results, Table 11 presents results analogous to Table 5 but replaces the mean HFCS index with the macroeconomic property price index. To ease the exposition, for each country, the table reports the household total net wealth difference to Germany (IHS transformed), the contribution of the explained part to this difference as well as the contributions of the four variables of main interest. The results in this table generally reinforce previously discussed results.

With regard to the *Oaxaca-Blinder decomposition*, replacing the mean HVA index with the macroeconomic index results in similar (IHS transformed) mean net wealth differences to Germany. Noteworthy, differences now are not statistically significant for Austria, Cyprus, Greece, the Netherlands and Slovakia. The total explained part turns insignificant for Cyprus, the Netherlands and Portugal. This is mainly due to the smaller sample size for these countries when using the macroeconomic property price index. With regard to the *RIF-OB decomposition*, replacing the mean HVA index with its macroeconomic index produces sizable variations for some selected coefficient estimates (Table 11). For example, at the median, this is particularly the case for the total difference and the explained part with regard to Austria, Cyprus, Greece and Slovakia.

Furthermore, with exception of Cyprus, the differences in homeownership rates always help to explain net wealth differences to Germany at the mean and median. Even at the 75th percentile homeownership contributes significantly to the explained part in 10 out of 12 countries. At 90th percentile the contribution remains significant for one half of our countries. Differences in property price developments contribute to the mean net wealth difference between Germany and five other countries (Spain, France, Greece, Italy and Portugal). For the median decomposition, it contributes significantly to the explained part in 9 out of 12 countries, at the 75th percentile decomposition in 7 countries and at the 90th percentile decomposition in 5 countries. In summary, main results are robust when we use the macro index.

Table 12 presents non-bootstrapped results of the *Oaxaca-Blinder decomposition* for all four property price indices. The three HVA indices show roughly similar contributions. Larger differences to the macroeconomic index are only visible for some countries. Homeownership is significant in all countries and specifications with the exception of Austria and to a minor extend Cyprus. The role of differences in intergenerational transfers and the years since the largest transfer seems equally economically important in all specifications, again with the exception of Cyprus and to smaller extent to Austria.

Table 13, Table 14 and Table 15 show the robustness of the results substituting the mean HVA index with the other HVA indices for different percentiles at the median and the upper end of the net wealth distribution (results non-bootstrapped). Still, the alteration of the index does not change the economic interpretation of the analysis in any of the analysed countries.

6 Conclusion

Recent results from the Eurosystem Household Finance and Consumption Survey reveal large wealth differences within the euro area, which has sparked a lot of media attention. This paper aims to uncover some of the main factors driving these differences. Doing so, we focus on three main factors: 1) homeownership, 2) property price dynamics and 3) intergenerational transfers. In the household wealth literature, these factors are recurringly found to be of importance for the wealth accumulation process, and differences across countries therein are therefore expected to contribute to explaining the observed wealth differences across euro area countries. For example, homeownership, which is the most important household asset, varies greatly in the euro area (44%-90%). Similarly, past house price dynamics in the last 20+ years differ substantially across euro area countries. In some countries, notably Germany, house prices have increased very modestly in the last twenty years (until 2010), whereas in other countries house price developments were very dynamic. In the latter countries, early investors into the household main residence were therefore able to benefit substantially. In this context, intergenerational transfers increase household wealth directly, as they are an important factor for the tenancy choice, i.e. for the decision to own or rent.

We show that these three factors, in addition to the common household and demographic factors, are relevant determinants of the household wealth accumulation process across euro area countries. Furthermore, using various econometric regressions and decomposition techniques, we show that (after controlling for household heterogeneity) homeownership and house price dynamics are important for explaining the observed wealth differences across euro area countries, whereas intergenerational transfers and gifts matter to a

much smaller extent. Homeownership and house price dynamics are particularly relevant contributing factors in middle part of the wealth distribution. Across euro area countries, these factors explain 56% of the difference in (IHS transformed) total net household wealth at their respective median level relative to Germany. The absolute contribution of these factors in explaining net wealth differences remains substantial but diminishes for higher percentiles of the net wealth distribution.

7 References

Andrews, D. and A. Caldera Sánchez (2011): The evolution of homeownership rates in selected OECD countries: Demographic and public policy influences. *OECD Journal: Economic Studies* **2011**: 207-243.

Andrews, D., A. Caldera Sánchez and Å. Johansson (2011): Housing markets and structural policies in OECD countries. OECD Economics Department Working Papers No. 836.

Barsky, R., J. Bound, K.C. Kerwin, and J.P. Lupton (2002): Accounting for the Black-White wealth gap: A nonparametric approach. *Journal of the American Statistical Association* **97**: 663–73.

Bauer, T. K., D.A. Cobb-Clark, V.A. Hildebrand and M. Sinning (2011): A comparative analysis of the nativity wealth gap. *Economic Inquiry* **49**(4): 989-1007.

Benítez-Silva, H.S. Eren, F. Heiland and S. Jiménez-Martin (2009): How well do individuals predict the selling prices of their homes? Working Paper No. 571. The Levy Economics Institute of Bard College.

Blau, F.D. and J.W. Graham (1990): Black-white differences in wealth and asset composition. *Quarterly Journal of Economics* **105**(2): 321-339.

Blinder, A.S. (1973): Wage discrimination: Reduced form and structural estimates. *Journal of Human Resources* 8(4): 436–455.

Bucks, B. and K. Pence (2008): Do homeowners know their house values and mortgage terms? *Journal of Urban Economics* **64**(2): 218-233.

Cameron, A.C. and P.K. Trivedi (2010): *Microeconometrics using Stata*. Revised edition, Texas: Stata Press.

Case, B. and S. Wachter (2005): Residential real estate price indices as financial soundness indicators: methodological issues, in Bank of International Settlements (eds.) *Real estate in- dicators and financial stability: Proceedings of a joint conference organised by the BIS and the IMF in Washington DC*, 27–28 October 2003. BIS Papers 21.

Chiuri, M.C. and T. Jappelli (2003): Financial market imperfections and home ownership: A comparative study. *European Economic Review* **47**: 857-875.

Christelis, D., D. Georgarakos and M. Haliassos (2013): Differences in portfolios across countries: Economic environment versus household characteristics. *Review of Economics and Statistics* **95**(1): 220–236.

Cobb-Clark, D.A. and V.A. Hildebrand (2006): The wealth of Mexican Americans. *Journal of Human Resources* **41**(4): 841-868.

Davies, J.S., S. Sandström, A.B. Shorrocks and E.N. Wolff (2011): The level and distribution of global household wealth. *Economic Journal* **121**: 223–254.

Deutsche Bundesbank (2013): Household wealth and finances in Germany: Results of the Bundesbank survey. *Monthly Report*, June 2013, 24-49.

Di Nardo, J., N. Fortin, and T. Lemieux (1996): Labor market institutions and the distribution of wages, 1973-1992: A semiparametric approach. *Econometrica* 64: 1001-1044.

European Central Bank (2009): Housing finance in the euro area. *Structural Issues Report*. Frankfurt: ECB.

Faiella, I. (2010): The use of survey weights in regression analysis, Bank of Italy Working Paper No. 739.

Fessler, P., P. Mooslechner and M. Schürz (2008): How inheritances relate to wealth distribution? Theoretical reasoning and empirical evidence on the basis of LWS data, Luxembourg Wealth Study Working Paper No. 6.

Firpo, S., N.M. Fortin and T. Lemieux (2007): Decomposing wage distributions using recentered influence functions regressions. Mimeo, University of British Columbia.

Firpo, S., N.M. Fortin and T. Lemieux (2009): Unconditional quantile regressions. *Econometrica* 77(3): 953-973.

Fortin, N.M., T. Lemieux and S. Firpo (2011): Decomposition methods in economics. In Ashenfelter, O. and D. Card (eds). *Handbook of Labor Economics* **4A**: 1–102, Amsterdam: Elsevier.

Gale, W.G. and K.M. Pence (2006): Are successive generations getting wealthier, and if so, why? Evidence from the 1990s. *Brookings Papers on Economic Activity* **1**: 155-213.

Goodman, J.L. and J.B. Ittner (1992): The accuracy of home owner's estimates of house value. *Journal of Housing Economics* **2**: 339–357.

Hilbers, P., A.W. Hoffmaister, A. Banerji and H. Shi (2008): House price developments in Europe: A comparison. IMF Working Paper 08/211.

Household Finance and Consumption Network (2013a): The Eurosystem Household Finance and Consumption Survey – Results from the first wave, ECB Statistics Paper No. 2.

Household Finance and Consumption Network (2013b): The Eurosystem Household Finance and Consumption Survey – Methodological report for the first wave, ECB Statistics Paper No. 1.

Ihlanfeldt, K.R. and J. Martinez-Vazquez (1986): Alternative value estimates of owneroccupied housing: Evidence on sample selection bias and systematic errors. *Journal of Urban Economics* **20**: 356–369.

Jann, B. (2008): The Blinder–Oaxaca decomposition for linear regression models. *Stata Journal* 8(4): 453–479.

Juhn, C., K. M. Murphy and B. Pierce (1993): Wage inequality and the rise in returns to skill, *Journal of Political Economy* **101**: 410-442.

Machado, J.F. and J. Mata (2005): Counterfactual decomposition of changes in wage distributions using quantile regression. *Journal of Applied Econometrics* **20**: 445-465.

Magee L., A.L. Robb and J.B. Burbidge (1998): On the use of sampling weights when estimating regression models with survey data. *Journal of Econometrics* **84**, 251-271.

Mathä, T.Y., A. Porpiglia and M. Ziegelmeyer (2014): Wealth differences across borders and the effect of real estate price dynamics: Evidence from two household surveys, BCL Working Paper 90 and ECB Working Paper 1672.

Oaxaca, R. (1973): Male-female wage differentials in urban labor markets. *International Economic Review* 14, 693-709.

Pence, K.M. (2006): The role of wealth transformations: An application to estimating the effect of tax incentives on saving. *B.E. Journals in Economic Analysis and Policy: Contributions to Economic Analysis and Policy* **5**(1): 1–24.

Piketty, T. (2011): On the long-run evolution of inheritance: France 1820-2050. *Quarterly Journal of Economics* **126**, 1071–1131.

Rao, J.N.K and C.F.J. Wu (1988): Resampling inference with complex survey data. *Journal* of the American Statistical Association 83: 231–241.

Rubin, D.B. (1987). *Multiple Imputation for Nonresponse in Surveys*. New York: John Wiley Sons.
Sinning, M. (2007): Wealth and asset holdings of immigrants in Germany. IZA DP No. 3089.

Wolff, E.N. and M. Gittleman (2011): Inheritances and the distribution of wealth or whatever happened to the great inheritance boom? BLS Working paper 445.

Appendix A: The Eurosystem Household Finance and Consumption Survey

The Eurosystem HFCS, coordinated by the ECB and implemented by 15 National Central Banks in the Eurosystem (with the exception of Ireland and Estonia), is an ex-ante harmonised micro database focussing on households' balance sheets and economic behaviour. It includes 62,521 observations representing 138,122,237 private households resident in the participating countries (HFCN, 2013a; HFCN, 2013b).

	Table	6: Survey Characte	eristics	
	Fieldwork	Assets & Liabilities	Income	Interview mode
AT	09/10 - 05/11	Time of interview	2009	CAPI
BE	04/10 - 11/10	Time of interview	2009	CAPI
CY	04/10 - 01/11	Time of interview	2009	88% PAPI, 12% CAPI
DE	09/10 - 07/11	Time of interview	2009	CAPI
ES	11/08 – 07/09	Time of interview	2007	CAPI
FI	01/10 - 05/10	31.12.2009	2009	97% CATI. 3% CAPI
FR	10/09 – 02/10	Time of interview	2009	CAPI
GR	06/09 – 09/09	Time of interview	Last 12 months	CAPI
IT	01/11 – 08/11	31.12.2010	2010	85% CAPI, 15% PAPI
LU	09/10 - 04/11	Time of interview	2009	CAPI
MT	10/10 - 02/11	Time of interview	Last 12 months	81% CAPI, 19 PAPI
NL	04/10 - 12/10	31.12.2009	2009	CAWI
PT	04/10 - 07/10	Time of interview	2009	CAPI
SI	10/10 – 12/10	Time of interview	2009	CAPI
SK	09/10 - 10/10	Time of interview	Last 12 months	CAPI

Note: CAPI (Computer Assisted Personal Interview), CATI (Computer Assisted Telephonic Interview, CAWI (Computer Assisted Web Interview), PAPI (Paper Assisted Personal Interview). Source: HFCN (2013b).

Table 6 highlights the main characteristics of the different country modules in terms of fieldwork period, reference year and survey mode. The reference year of the most surveys is 2010 (2009 for income). The data collection period is not fully harmonised (spanning from November 2008 (Spain) to August 2011 (Italy)). Still, any resulting biases are expected to be limited. For the same reason we chose not to apply any PPP correction mechanism (for detailed discussion of comparability of country surveys, see HFCN, 2031b). Table 7 shows that the predominant share of interviews were conducted by CAPI (Computer Assisted Personal Interviews). In some countries CAPI was supplemented with PAPI (Paper Assisted Personal Interview). CAWI (Computer Assisted Web Interview) is exclusively used by the Dutch survey. The sampling strategy mainly relies on stratified random sampling (with the exception of Slovakia) and reflects country specificities. In addition, some countries apply oversampling of wealthy household; this is mainly due the reduction of variance in the main wealth components. Each country module is intended to

be country representative (with the exception of Slovenia). Allowing for different sampling strategies, the weighting procedure follows a harmonised method (HFCN, 2013b).

Country	Number of households	Number of obs	Type of sampling design	Oversampling of wealthy
Euro Area	138,122,237	62,521	-	-
AT	3,773,956	2,380	2 stage stratified sampling	yes
BE	4,692,601	2,327	1 stage stratified sampling	yes
CY	303,242	1,237	1 stage stratified sampling	yes
DE	39,673,000	3,565	3 stage stratified sampling**	yes
ES	17,017,706	6,197	2 stage stratified sampling*	yes
FI	2,531,500	10,989	1 stage stratified sampling	yes
FR	27,860,408	15,006	2 stage stratified sampling	yes
GR	4,114,150	2,971	2 stage stratified sampling	yes
IT	23,817,962	7,951	2 stage stratified sampling	no
LU	186,440	950	1 stage stratified sampling	yes
MT	143,677	843	1 stage stratified sampling	no
NL	7,386,144	1,301	1 stage stratified sampling	no
PT	3,932,010	4,404	2 stage stratified sampling	yes
SI	777,777	343	2 stage stratified sampling	yes
SK	1,911,664	2,057	1 stage stratified quota sampling	no

Table 7: Sampling strategy

Notes: * In Spain, one stage for households living in municipalities with over 100,000 inhabitants, two stages for others. ** In Germany, three stages for households living in municipalities with over 100,000 inhabitants, two stages for others. Source: HFCN (2013b).

Logical inconsistencies of replies and item non-response are addressed via systematic editing and multiple imputation processes (based on 5 implicates) (Rubin, 1987), using state of art methodologies and committing to broad covariate conditioning. This procedure ensures unbiased results and preserves the covariance structure of the questionnaire. In addition, to allow a correct variance estimation of each point estimate, 1000 replicate weights are calculated following the Rao-Wu method (Rao and Wu, 1988) and taking into account the sample design, unit-non-response and additional calibration information (see HFCN, 2013b for further details and a list of the imputed variables in each country module).

Appendix B: Variable definitions and summary statistics

Variable name	Variable description
total net wealth	total net wealth is the difference between total gross assets (real and financial
	assets) and total liabilities as defined in Annex I of HFCN (2013a)
Main household character	istics
male (d)	reference person is male
age, age2	age and age squared of the reference person
single (d)	reference person is single
married (d) (ref.)	reference person is married or has a consensual union on a legal basis
divorced (d)	reference person is divorced
widowed (d)	reference person is widowed
hhsize	number of household members
lowedu (d) (ref.)	reference person with low education (ISCED=0,1,2)
midedu (d)	reference person with medium education (ISCED=3,4)
higheduc (d)	reference person with high education (ISCED=5,6)
born in country of resi-	reference person is born in the country of residence
dence (d)	1
Employment and income r	elated characteristics
ihs(total income)	inverse hyperbolic sine transformation of total household gross income (as de-
、	fined in Annex I of HFCN, 2013a) in log form
temporary contract (d)	reference person has a temporary working contract
employee (d) (ref.)	main labour status of reference person is employee
self-employed (d)	main labour status of reference person is self-employed
unemployed (d)	main labour status of reference person is unemployed
retired (d)	main labour status of reference person is retired
other (d)	main labour status of reference person is other empl. status not listed before
employment status miss-	
ing (d)	
financial sector (d)	reference person works in the financial sector (NACE: K)
public sector (d)	reference person works in the public sector (NACE: O, P, Q)
Variables related to home	ownership and house prices
owner	household owns fully or partially the HMR
macro index	macroeconomic property price index
mean HVA index	smoothed mean housing value appreciation index
median HVA index	smoothed median housing value appreciation index
median ratio HVA index	smoothed median ratio housing value appreciation index
Variables related to interg	
	inverse hyperbolic sine transformation of the total amount of received gifts or
fers)	inheritances in log form (at the time of transfer; including HMR)
years since largest trans-	number of years since the largest gift or inheritance was received
fer	
Country fixed effects	
country (d)	C=1 if household resident in C \in {AT, BE, CY, ES, FR, GR, IT, LU, MT, NL, PT,
- · ·	SL}, zero otherwise; DE is reference country.

(d) denotes variable being a dummy variable. (ref.) indicates the reference group in the estimations.

Basic summary statistics

Variable	Obs	Mean	Std. Err.	Min	Max
AT	62,521	2.7%	0.065%	0	1
BE	62,521	3.4%	0.072%	0	1
CY	62,521	0.2%	0.019%	0	1
ES	62,521	12.3%	0.131%	0	1
FR	62,521	20.2%	0.160%	0	1
GR	62,521	3.0%	0.068%	0	1
IT	62,521	17.2%	0.151%	0	1
LU	62,521	0.1%	0.015%	0	1
MT	62,521	0.1%	0.013%	0	1
NL	62,521	5.3%	0.090%	0	1
PT	62,521	2.8%	0.067%	0	1
SI	62,521	0.6%	0.030%	0	1
SK	62,521	1.4%	0.047%	0	1
male	62,521	54.3%	0.199%	0	1
age	62,521	53	0.069	16	100
age2	62,521	3074	7.526	256	10000
single	62,514	22.2%	0.166%	0	1
divorced	62,514	10.7%	0.124%	0	1
widowed	62,514	13.4%	0.137%	0	1
hhsize	62,521	2.32	0.005	1	16
mideduc	62,370	41.4%	0.197%	0	1
higheduc	62,370	23.6%	0.170%	0	1
born in country of residence	40,017	89.1%	0.155%	0	1
ihs(total income)	62,521	10.4	0.005	-13.1	16.0
self-employed	62,521	8.2%	0.110%	0	1
unemployed	62,521	5.4%	0.090%	0	1
retired	62,521	30.9%	0.185%	0	1
other	62,521	10.3%	0.121%	0	1
employment status missing	62,521	0.8%	0.035%	0	1
temporary contract	57,930	5.3%	0.108%	0	1
financial sector	62,241	2.0%	0.057%	0	1
public sector	62,241	12.3%	0.140%	0	1
owner	62,521	60.1%	0.196%	0	1
macro index	40,385	2.1	0.030	0	52.7
mean HVA index	33,478	2.0	0.019	0	27.9
median HVA index	33,478	2.4	0.027	0	46.0
median ratio HVA index	33,478	2.4	0.027	0	46.0
ihs(amount gifts & transfers)	43,063	3.4	0.024	0	17.9
years since largest transfer	43,105	5.1	0.053	0	86

Source: own calculations based on the HFCS UDB 1.0; data are multiply imputed and weighted.



Figure 10: Construction of mean HVA index over countries

Appendix C: Construction of HVA indices

Source: own calculations based on the HFCS UDB 1.0; data are multiply imputed and weighted. Slovenia is excluded since it has no country representative survey. Information on the initial value of the HMR does not exist for Finland and France. The kernel-weighted local polynomial regression uses the Epanechnikov kernel-function. The kernel bandwidth is 5 and the degree of the polynomial smooth is 1. The confidence intervals are determined by the kernel smoothing.

year	AT	BE	CY	DE	ES	GR	IT	LU	МТ	NL	PT	SI	SK
1925			1.0	1.0	4.0	1.0		2.0	1.0				
1927					1.0	1.0							
1929					1.0						1.0		
1930					1.0	1.2	1.0				2.0		
1932	1.0				1.0								
1933			1.0		1.0				1.0		1.0	1.0	
1934					1.0						1.0		
1935			1.0		2.0	1.0	1.0				1.0		
1936					2.0	1.2		1.0			1.0	1.0	
1937	1.0				1.0	1.2		1.0			1.0	1.0	
1938	0.2			1.2	2.0			1.0					
1939	0.2	1.0	1.0	1.2	2.0	1.0	1.0	1.0					
1940	1.0	1.0	1.0		7.0	1.0	1.0				4.0		
1940	1.0	1.0		1.2	2.2		1.0				4.0		
1941				1.2			1.0				2.0		
					4.0		1.0				2.0		
1943			0.2		4.0						1.0		
1944					3.0	3.0		0.4			3.0		
1945	1.2		1.0		9.0	3.0	2.0		2.0		5.0		1
1946		1.0			2.0	1.0					2.0		
1947	0.2				4.0	1.4	4.0	1.0			2.0		
1948	1.0				4.2		4.0						
1949	1.2			0.2	3.0	3.0	4.0				2.0		
1950	8.0	1.0	3.0	3.4	30.0	23.2	19.0	1.2	1.0		26.0	1.0	3
1951	3.0	2.0	1.0	4.0	5.0	4.2	2.0		1.0		2.0		
1952	1.2	2.0	0.2	3.2	10.8	6.0	1.0		1.0		1.0	1.0	
1953	1.0	3.2		2.0	5.4	6.2	5.0	2.0		1.0	3.2	1.0	1
1954	1.0	4.0	2.0	4.0	13.0	5.2	6.0				7.0	1.0	
1955	6.0	4.0	4.0	2.0	14.6	6.0	14.0	3.2	6.0		17.4	3.0	1
1956	2.0	9.0		3.2	12.2	2.2	13.0	1.0	0.0	1.0	7.4	5.0	1
1957	2.0	4.2	1.0	5.2	7.2	4.0	14.0	1.0	3.0	1.0	9.0		1
1958	4.6	14.6	1.0	5.2	24.8	5.2	18.0	1.0	1.0	1.0	9.4		
1959	5.2	14.0		7.0	24.8	11.0	17.0	0.2	2.0	1.0	4.4	1.0	1
			1.0							4.0		1.0	
1960	13.2	13.0	1.0	10.0	86.2	53.8	73.0	4.0	7.0	4.0	37.6	2.0	18
1961	2.0	8.2	2.0	10.4	19.4	6.0	22.0	1.0	1.0	2.0	10.8	2.0	2
1962	8.4	13.0	2.0	7.0	41.4	6.0	26.0	3.0	3.0		11.6	3.0	6
1963	9.0	18.2	5.0	5.0	42.6	8.2	25.0	3.0	3.0	2.0	14.0	3.0	1
1964	9.0	12.4	1.0	11.0	43.8	17.0	33.0	3.0	4.0	2.0	11.0	2.0	1
1965	19.2	19.0	4.0	19.0	56.0	29.4	50.0	2.0	5.0	3.0	34.4		9
1966	7.2	15.4	2.0	14.0	52.8	10.4	27.0	3.0	1.0	6.0	23.4	3.0	4
1967	10.2	23.0	5.0	23.4	42.8	23.8	40.0	3.0	4.0	5.0	20.2	2.0	2
1968	15.8	21.2	8.0	26.0	77.4	16.4	56.0	3.0	1.0	4.0	23.0	2.0	7
1969	7.8	22.0	1.0	16.0	78.8	24.8	53.0	7.0	2.0	1.0	22.0	3.0	3
1970	31.4	29.6	9.2	36.0	150.0	57.0	131.0	9.0	13.0	4.0	74.2	7.0	35
1971	9.6	14.0	2.2	25.0	52.4	8.8	54.0	10.0	3.0	5.0	17.4	4.0	1
1972	12.4	20.2	3.2	35.0	97.0	20.6	61.0	6.2	7.0	11.0	33.6	4.0	6
1973	18.4	27.0	6.2	28.2	127.4	14.4	53.0	8.0	7.0	16.0	39.2	5.0	11
1974	20.2	24.6	4.0	29.2	93.8	33.2	56.0	13.0	7.0	9.0	58.2	2.0	10
1975	24.6	43.2	9.4	29.4	133.6	33.0	70.0	9.0	17.0	9.0	75.4	4.0	25
1976	16.6	36.2	5.4	43.2	88.8	18.8	63.0	11.2	14.0	17.0	57.2	1.0	4
1977	12.8	27.0	11.2	33.4	71.0	20.8	45.0	7.0	8.0	10.0	42.8	9.0	e
1978	19.4	33.4	12.2	41.0	147.4	28.8	98.0	11.0	16.0	12.0	51.8	3.2	14
1979	19.4	33.4	4.2	36.0	118.0	60.2	55.0	12.0	20.0	8.0	40.6	6.0	8
1980	56.6	52.0	25.0	60.2	171.4	111.6	186.0	12.2	31.0	17.0	173.6	11.0	57
1981	19.4	32.2	11.2	36.2	84.0	28.4	60.0	9.0	15.0	15.0	33.6	4.0	10
1982	24.6	30.0	15.4	32.4	108.8	29.6	77.0	10.2	24.0	14.0	75.4	9.2	12
1983	31.4	22.2	16.2	29.0	118.2	29.2	84.0	9.0	21.0	15.0	35.0	3.4	11
1984	20.4	32.0	11.4	33.6	126.0	45.2	87.0	8.0	10.0	17.0	57.4	6.0	18
1985	36.4	44.0	27.0	47.0	132.4	42.2	136.0	12.2	19.0	26.0	110.6	11.2	43
1986	16.6	22.2	19.0	48.6	91.0	21.2	84.0	10.0	21.0	23.0	54.6	6.4	11
1987	17.8	34.4	14.2	32.4	95.0	24.0	81.0	8.0	16.0	18.0	58.0	4.0	14
1988	22.2	34.2	17.0	26.0	151.8	22.4	77.0	13.0	18.0	28.0	70.2	5.6	25
1989	17.6	33.2	15.0	49.2	139.4	58.6	65.0	17.0	21.0	35.0	50.0	10.2	15
1990	50.6	42.0	28.0	51.6	163.0	92.8	163.0	14.0	34.0	26.0	125.0	18.6	80
1991	26.8	35.4	13.8	40.4	81.6	33.4	87.0	17.0	12.0	24.0	41.8	10.6	18
1992	20.4	45.2	18.2	51.6	134.4	32.0	63.0	19.0	27.0	26.0	70.0	6.0	31
1993	17.0	42.2	21.0	47.0	101.2	26.4	49.0	13.0	14.0	32.0	68.2	7.6	29
1994	16.8	29.6	20.0	58.8	147.4	46.0	83.0	16.0	12.0	17.0	71.4	4.2	25
1995	42.6	54.0	25.0		118.6	37.4	85.0	18.0	16.0	28.0	108.0	7.2	
1996	15.8	38.0	19.0			29.2		18.0	21.0		71.8	6.0	
1996 1997	15.8	38.0 44.0			121.4		68.0 81.0		20.0	35.0 29.0	65.2	6.0 7.0	30
			17.0					11.6					
1998	32.2	54.0	16.0		185.8	59.0	98.0	17.0	15.0	24.0	136.8	6.2	53
1999	30.2	42.2	18.0	57.6	170.6	81.2	87.0	21.0	11.0	39.0	142.0	3.2	44
2000	39.2	51.0	52.0	59.8	176.4	95.2	152.0	20.0	23.0	26.0	146.6	13.2	145
2001	36.8	35.0	21.0	60.4	99.6	25.2		12.0		31.0	50.4	7.0	56
2002	19.6	36.0	35.0	56.6	106.4	33.2	100.0	28.0	18.0	18.0	55.2	2.0	57
2003	19.6	44.2	43.0		117.6	43.4	75.0	18.0	17.0	27.0	36.2	6.0	71
2004	39.0	41.0	49.0		115.4	65.0	91.0	31.0	7.0	31.0	55.2	5.0	72
	40.2	60.0	69.6	64.0	95.2		109.0	25.0	5.0	44.0	71.2		104
2005	26.4	42.0	62.4		106.6	52.8	80.0	20.2	5.0	41.0	59.0		67
2005 2006						62.2	80.0	33.2	1.0	44.0			
2006		49.0	80.0	54.0									
2006 2007	33.8	49.0 39.2	80.0 58.0		74.6 51.8						49.0 48.2	4.0 4.0	
2006 2007 2008	33.8 27.2	39.2	58.0	49.2	51.8	48.2	66.0	23.0	3.0	39.0	48.2	4.0	80 66 40
2006	33.8												

Table 8: Number of observations for index construction

ber over 5 implicates is shown. Information on the initial value of the HMR does not exist for Finland and France.



Figure 11: Comparison of macroeconomic indices and HVA indices with and without adjustment for the size of the HMR

Source: own calculations based on the HFCS UDB 1.0; data are multiply imputed and weighted. Slovenia is excluded since it has no country representative survey. Information on the initial value of the HMR does not exist for Finland and France. The square meter size of the HMR is not available in Malta and Portugal. The kernel-weighted local polynomial regression uses the Epanechnikov kernel-function. The kernel bandwidth is 5 and the degree of the polynomial smooth is 1.

ılts
resu
ness
usti
Rok
Ö×
endi
App

			I	Iable	e y: Mec	uan reg	I able 9: Median regression:	·					
			- D	ifferent p	roperty p	- Different property price index specifications	c specifica	ttions -					
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)
	AT	BE	CY	DE	ES	FR	GR	IT	ΓΩ	MT	N	PT	SK
macro index													
owner	1.841 ***	2.469 ***	1.664 **	1.104 ***	2.723 ***	2.265 ***	2.288 ***	2.568 ***	1.984 ***	2.076 ***	1.785 ***	2.375 ***	1.983 ***
	(0.185)	(0.104)	(0.821)	(0.140)	(0.130)	(0.042)	(0.110)	(0.034)	(0.175)	(0.186)	(0.194)	(0.114)	(0.223)
macro index	0.109	0.029 **	0.064	0.532 ***	0.012 ***	0.037 ***	0.165 ***	0.005 ***	0.040 **	0.034	0.079 ***	0.211 ***	0.286 *
	(0.100)	(0.012)	(0.743)	(0.099)	(0.004)	(0.008)	(0.035)	(0.001)	(0.018)	(0.021)	(0.021)	(0.046)	(0.164)
ihs(amount gifts & transfers)	0.056 ***	0.029 ***	0.079 ***	0.059 ***	0.039 ***	0.053 ***	0.040 ***		0.029 ***	0.034 ***	0.027 **	0.052 ***	0.031 ***
	(0.008)	(0.005)	(0.021)	(0.005)	(0.004)	(0.002)	(0.008)		(0.007)	(0.007)	(0.013)	(0.007)	(0.010)
years since largest transfer	-0.000	0.005 *	-0.016	-0.010 ***	0.002	-0.000	-0.028 ***		-0.007	-0.004	-0.000	-0.007	-0.002
mean HVA index (reference specification)	e specificati	(000)	(/10:0)	(000.0)	(00010)	(100.0)	(2007)		(000.0)	(con.n)	(600.01	(000/01)	(#10.01
	1 OTT 444	10, 10, 144		1 100 444	0101			0 401 444	1 011 444	777 LOO 0	1 007 444	*** FOL 0	244 1444
owner	1.8/2	2.426	7.56.2	1.189 ***	7.781		7.5/4	2.48/ ***	1.900 T		1.896 ***	7.724 ***	2.446 ***
	(0.184)	(0.107) 0.0FF ***	0.283)	0.129)	(0.130)		(0.096)	(0.037)	(0.177)	(0.170)	(0.191)	0.094)	0.106)
mean HVA index	0.133 *	1.0.0	1 50.0	0.438 ***			0.070 ***	0.027 ***		0.019	0.043 *	0.035 ***	0.056 ***
	(0.081)	(0.017)	(0.024)	(0.074)	0.002)		(0.013)	(0.004)	(0.027)	(0.012)	(0.026)	(0.006)	(0.008)
ihs(amount gifts & transfers)	0.054 ***		800.0	*** 650.0	0.039		0.033 ***		0.030 ***	0.036 ***	0.033 **	0.043 ***	0.030 ***
	(0.007)	(0.005)	(0.010)	(0.004)	(0.003)		(0.005)		(0.007)	(0.007)	(0.014)	(0.005)	(0.005)
years since largest transfer	-0.006 *	0.002	-0.007	-0.010 ***	-0.001		-0.017 ***		-0.005	-0.006	-0.004	-0.008 ***	-0.011 ***
	(0.003)	(0.003)	(0.005)	(0.002)	(0.002)		(0.002)		(0.005)	(0.007)	(0.011)	(0.003)	(0.003)
median HVA index													
owner	1.849 ***	2.450 ***	2.550 ***	1.294 ***	2.784 ***		2.669 ***	2.519 ***	1.937 ***	2.107 ***	1.890 ***	2.792 ***	2.454 ***
	(0.197)	(0.106)	(0.277)	(0.123)	(0.130)		(0.092)	(0.036)	(0.177)	(0.164)	(0.190)	(0.091)	(0.101)
median HVA index	0.147 *	0.042 ***	0.048 **	0.353 ***	0.002		0.024 ***	0.016 ***	0.104 ***	0.015	0.041 *	0.010 ***	0.044 ***
	(0.083)	(0.014)	(0.022)	(0.074)	(100.0)		(0.005)	(0.002)	(0.024)	(0.010)	(0.024)	(0.002)	(0.006)
ihs(amount gifts & transfers)	0.055 ***	0.029 ***	0.058 ***	0.059 ***	0.039 ***		0.030 ***		0.031 ***	0.035 ***	0.033 **	0.041 ***	0.030 ***
	(0.007)	(0.005)	(0.010)	(0.004)	(0.003)		(0.005)		(0.007)	(0.007)	(0.014)	(0.005)	(0.005)
years since largest transfer	-0.006 **	0.002	-0.007	-0.011 ***	-0.000		-0.015 ***		-0.006	-0.005	-0.004	-0.008 **	-0.011 ***
median ratio HVA index	(000-00)	(0000)	Connect	(00010)	(20010)		(-0010)		(00010)	(20010)	(*** 010)	(mont)	(0000)
	1 016 ***	*** VJV C	*** C75 C	1 267 ***	0 70E ***		*** CV7 c	0 510 ***	1 070 ***	0 1 AO ***	1 201 ***	1 702 ***	*** LJV C
TOTAC	0.157)	10101	20012	(0 136)	0.130)		() (031)	0.036)	(0.176)	0.1460)	1.0.1	(0000)	10105)
median ratio HVA index	0.121 *	0.040 ***	0.042 **	0.384 ***	0.002		0.038 ***	0.016 ***	0.092 ***	0.010	0.039 *	0.010 ***	0.043 ***
	(0.073)	(0.014)	(0.020)	(0.080)	(0.001)		(0.008)	(0.002)	(0.023)	(0.00)	(0.023)	(0.003)	(0.006)
ihs(amount gifts & transfers)	0.053 ***	0.029 ***	0.058 ***	0.059 ***	0.039 ***		0.031 ***		0.031 ***	0.034 ***	0.033 **	0.041 ***	0.030 ***
	(0.006)	(0.005)	(0.010)	(0.004)	(0.003)		(0.005)		(0.007)	(0.008)	(0.014)	(0.005)	(0.005)
years since largest transfer	-0.006 *	0.002	-0.007	-0.011 ***	-0.000		-0.016 ***		-0.006	-0.005	-0.004	-0.008 ***	-0.011 ***
	(0.003)	(0.003)	(0.006)	(0.002)	(0.002)		(0.002)		(0.005)	(0.007)	(0.011)	(0.003)	(0.003)
in all specifications													
socio-demographics	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
employment controls	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes	yes
min. number of observations	IS												
specification macro index	1829	1894	418	3226	4355	12197	1720	7378	868	674	1182	2899	851

Table 9: Median regression:

Source: own calculations based on the HFCS UDB 1.0; results adjusted for multiple imputation; robust standard errors in parentheses; * p < 0.1 ** p < 0.05 *** p < 0.01. All monetary values (total net wealth, total income and intergenerational transfers) are transformed using an inverse hyperbolic sine transformation in log form. specification HFCS indices

	Table . - <i>Diff</i>	10: Pooled n erent HVA an	Table 10: Pooled median regression for the euro area - Different HVA and macroeconomic index specifications -	ssion for the nic index speci	euro area fications -			
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
	all	ex FR	ex FR	ex FR	ex IT	ex IT & FR	ex IT & FR	ex IT & FR
owner	2.538 *** (0.018)	2.543 *** (0.019)	2.563 *** (0.019)	2.565 *** (0.018)	2.315 *** (0.024)	2.415 *** (0.024)	2.431 *** (0.023)	2.433 *** (0.023)
macro index	0.005 *** (0.001)				0.033 *** (0.005)			
mean HVA index		0.017 *** (0.002)				0.016 *** (0.003)		
median HVA index			0.009 *** (0.002)				0.008 *** (0.002)	
median ratio HVA index				0.009 *** (0.001)				0.008 *** (0.002)
ihs(amount gifts & transfers)					0.048 *** (0.002)	0.042 *** (0.002)	0.042 *** (0.002)	0.042 *** (0.002)
years since largest transfer					-0.001 (0.001)	-0.005 *** (0.001)	-0.005 *** (0.001)	-0.005 *** (0.001)
constant	3.125 *** (0.154)	4.200 *** (0.134)	4.173 *** (0.129)	4.167 *** (0.128)	3.512 *** (0.189)	4.249 *** (0.209)	4.224 *** (0.190)	4.222 *** (0.194)
socio-demographic variables employment controls final population weights country fixed effects	yes yes yes	yes yes yes	yes yes yes	yes yes yes	yes yes yes	yes yes yes	yes yes yes	yes yes yes
minimum number of observations	40019	33231	33231	33231	32195	25723	25723	25723

Source: own calculations based on the HFCS UDB 1.0; results adjusted for multiple imputation; robust standard errors in parentheses; *p<0.1 **p<0.05 ***p<0.01. All monetary values (total net wealth, total income and intergenerational transfers) are transformed using an inverse hyperbolic sine transformation in log form.

l 90th quantile	
75th and 90th	
and	
75th	
0th, 7	
50	,
mean,	
at	,
Table 11: OB and RIF-OB decomposition at mean, 50th, 75th and 90	
OB	
RIF-(
р	
3 aı	ł
Ю	
11:	
ole	
Tak	

		To	tal net wei	ulth, macr	Total net wealth, macroeconomic index	I	500 bootst	500 bootstrap replicates	ates				
		(1)	(2)	(3)	(1)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
		AT	BE	CY	ES	FR	GR	IT	ΓΩ	МТ	NL	ΡT	SK
	difference	0.32	1.83 ***	0.28	1.70 ***	1.07 ***	0.18	2.11 ***	2.22 ***	2.55 ***	-0.26	0.51 **	0.13
	explained	-0.42 ***	0.92 ***	0.00	1.66 ***	0.18 ***	0.39 **	0.74 ***	0.95 ***	0.81 ***	0.33	-0.21	0.86 **
	macro index	0.00	0.04	-0.02	0.17 ***	0.05 *	0.20 **	0.05 ***	0.10	0.02	0.18	0.23 ***	0.08
duu -və	owner	-0.15 ***	1.05 ***	0.00	1.73 ***	0.33 ***	0.31 ***	0.90 ***	0.90 ***	1.14 ***	0.34 ***	0.44 ***	0.76 ***
	ihs(gifts & transfers)	0.01	0.00	-0.17 **	-0.07 ***	0.02	-0.10 *		-0.03	-0.02	-0.31	-0.14 ***	-0.01
	years since largest transfer	-0.01	-0.01	0.07	0.00	0.00	0.11		0.02	0.02	0.04	0.01	-0.01
	difference	-0.16	1.44 ***	0.63 ***	1.36 ***	0.68 ***	0.37 ***	1.37 ***	2.02 ***	1.47 ***	0.78 ***	0.27 ***	-0.26 ***
юц	explained	-0.27 ***	0.54 ***	-0.08	0.28 ***	0.44 ***	0.39 ***	0.38 ***	0.66 ***	0.35 ***	0.48 ***	0.00	0.08
iso	macro index	0.00	0.19 ***	-0.01	0.11 ***	0.21 ***	0.18 ***	0.02 *	0.20 ***	0.24 ***	0.30 ***	0.13 ***	0.10
	owner	-0.19 ***	0.34 ***	0.03	0.32 ***	0.32 ***	0.25 ***	0.46 ***	0.35 ***	0.29 ***	0.22 ***	0.33 ***	0.37 ***
Ч1(02ә 8	ihs(gifts & transfers)	0.01	-0.01	-0.11 **	-0.03 ***	0.03	-0.10 ***		-0.02	-0.02	-0.07	-0.08 ***	-0.01
	years since largest transfer	0.00	0.00	0.09 **	0.01	-0.01	0.10 *		0.00	0.02 *	0.03	0.00	-0.02
	difference	0.02	0.73 ***	0.34 **	0.52 ***	0.31 ***	-0.11 *	0.54 ***	1.27 ***	0.68 ***	0.25 ***	-0.31 ***	-1.01 ***
юij	explained	-0.15	0.24 ***	0.01	0.03	0.11 ***	0.05	0.06 **	0.32 ***	0.07	0.13 *	-0.29 ***	0.03
isoc	macro index	-0.01	0.13 ***	-0.01	0.10 ***	0.14 ***	0.10 *	0.04 ***	0.17 ***	0.13	0.23 ***	0.06	0.10
Per Par Par Par Par Par Par Par Par Par Pa	owner	-0.11 **	0.09 ***	-0.04	0.10 ***	0.05 ***	0.06 **	0.17 ***	0.06 **	0.11 *	0.09 ***	0.14 ***	0.08
))) (ihs(gifts & transfers)	0.01	-0.01	-0.10	-0.03 ***	0.02	-0.08 **		-0.03	-0.02	-0.20 ***	-0.06 ***	-0.01
	years since largest transfer	-0.01	0.00	0.03	0.01 *	-0.01 *	0.06		0.01	0.01	0.08 *	0.00	0.05
	difference	0.15 **	0.50 ***	0.50 ***	0.35 ***	0.17 ***	-0.32 ***	0.34 ***	1.12 ***	0.50 ***	0.03	-0.39 ***	-1.15 ***
юц	explained	0.00	0.08 **	0.08	-0.07	0.01	-0.02	-0.01	0.21 **	0.05	-0.15 *	-0.38 ***	-0.25
isoc	macro index	0.00	0.02	0.00	0.11 ***	0.06 ***	0.08	0.04 ***	0.07	0.18	0.15 ***	* 60.0	0.20
Per Par Par Par Par Par Par Par Par Par Pa	owner	-0.07 **	0.05 **	0.05	0.04	0.03 ***	0.00	0.10 ***	0.06 *	-0.02	0.04 **	0.04	-0.11
))) (ihs(gifts & transfers)	0.01	0.00	-0.06	-0.05 ***	0.02	-0.05		-0.02	-0.02	-0.27 ***	-0.05 **	-0.01
	years since largest transfer	-0.01	0.00	-0.02	0.01	-0.01 *	0.01		0.00	-0.02	0.05	0.01	0.05
Source: ow *** p<0.01 form. Sloven employment	Source: own calculations based on the HFCS UDB 1.0; results adjusted for multiple imputation and bootstrapped standard errors with 500 replicates; * p<0.1 ** p<0.05 *** p<0.01. All monetary values (total net wealth, total income and intergenerational transfers) are transformed using an inverse hyperbolic sine transformation in log form. Slovenia is excluded since it has no country representative survey. Variables are grouped as follows: Employee status (self-employed, unemployed, retired, other employment status, tennorary contract. financial sector, public sector), democraphics (male, age, age2, single, divorced, widowed, histze, mideduc, hisheduc, born in	HFCS UDB , net wealth, i no country re financial se	1.0; results otal incom presentativ ctor, publi	adjusted fo e and inter re survey. " c sector), d	IDB 1.0; results adjusted for multiple imputation and bootstrapped standard errors with 500 replicates; * p<0.1 ** p<0.05 ulth, total income and intergenerational transfers) are transformed using an inverse hyperbolic sine transformation in log try representative survey. Variables are grouped as follows: Employee status (self-employed, unemployed, retired, othen ial sector, public sector), demographics (male, age, age2, single, divorced, widowed, hhsize, mideduc, hisheduc, born in	imputation al transfer re groupec cs (male, a	t and boots 's) are trar d as follow tge, age2,	trapped sta 1sformed ux s: Employe single, div	mdard erro sing an inw e status (se orced, wic	rs with 500 erse hyperi elf-employe lowed, hhs	9 replicates bolic sine t d, unempl size, midedi	;; * p<0.1 ransformat oyed, retir uc, highedu	** p<0.05 ion in log ed, other c. born in
the country	the country of residence). Household total net w	al net wealth	differences	are calcu	ealth differences are calculated relative to Germany	ve to Germ	any.	D				D	

			Table - Diffe	Table 12: Oaxaca-Blinder decomposition - Different HVA and macro index specifications -	aca-Blinc and macre	ler decon) index spe	mposition	с ⁻				
	(1)	(2)	(3)	(4)	(2)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
	AT	BE	CY	ES	FR	GR	IT	ΓΩ	ΤM	NL	ΡT	SK
Mean HVA index												
difference	0.73 ***	1.79 ***	2.06 ***	1.85 ***		1.15 ***	1.96 ***	2.21 ***	2.58 ***	-0.27	0.97 ***	1.54 ***
explained	0.02	0.91 ***	1.72 ***	1.76 ***		1.22 ***	0.64 ***	0.88 ***	0.86 ***	0.30	0.28 **	2.07 ***
mean HVA index	0.05	0.10	0.33 **	0.09 **		0.22 **	0.10 ***	0.13	-0.07	0.16	0.13 ***	0.18 ***
owner	0.08	0.99 ***	1.23 ***	1.87 ***		1.15 ***	0.77 ***	0.90 ***	1.23 ***	0.33 ***	0.92 ***	2.02 ***
ihs(gifts & transfers)	0.04 *	0.00	0.00	-0.06 ***		-0.03 *		-0.03	-0.01	-0.33	-0.11 ***	0.01
years since largest transfer	-0.03	-0.01	-0.02	0.00		-0.02		0.01	0.01	0.06	0.00	0.00
Macro index												
difference	0.32	1.83 ***	0.28	1.70 ***	1.07 ***	0.18	2.11 ***	2.22 ***	2.55 ***	-0.26	0.51 ***	0.13
explained	-0.42 ***	0.92 ***	0.00	1.66 ***	0.18 ***	0.39 **	0.74 ***	0.95 ***	0.81 ***	0.33	-0.21	0.86 **
macro index	0.00	0.04	-0.02	0.17 ***	0.05 *	0.20 **	0.05 ***	0.10	0.02	0.18	0.23 ***	0.08
owner	-0.15	1.05 ***	0.00	1.73 ***	0.33 ***	0.31 ***	0.90 ***	0.90 ***	1.14 ***	0.34 ***	0.44 ***	0.76 ***
ihs(gifts & transfers)	0.01	0.00	-0.17 **	-0.07 ***	0.02 **	-0.10 **		-0.03	-0.02	-0.31	-0.14 ***	-0.01
years since largest transfer	-0.01	-0.01	0.07	0.00	0.00	0.11		0.02	0.02	0.04	0.01	-0.01
Median HVA index												
difference	0.73 ***	1.79 ***	2.06 ***	1.85 ***		1.15 ***	1.96 ***	2.21 ***	2.58 ***	-0.27	0.97 ***	1.54 ***
explained	0.03	0.91 ***	1.71 ***	1.75 ***		1.19 ***	0.63 ***	0.88 ***	0.86 ***	0.29	0.27 **	2.08 ***
median HVA index	0.06	0.10	0.30 **	0.06		0.12 *	0.07 ***	0.14	-0.07	0.16	0.06 *	0.18 ***
owner	0.08	0.99 ***	1.25 ***	1.89 ***		1.20 ***	0.78 ***	0.90 ***	1.22 ***	0.33 ***	0.96 ***	2.03 ***
ihs(gifts & transfers)	0.04 *	0.00	0.00	-0.05 ***		-0.03 *		-0.03	-0.01	-0.33	-0.10 ***	0.01
years since largest transfer	-0.03	-0.01	-0.02	0.00		-0.01		0.01	0.01	0.06	0.00	0.00
Median ratio HVA index												
difference	0.73 ***	1.79 ***	2.06 ***	1.85 ***		1.15 ***	1.96 ***	2.21 ***	2.58 ***	-0.27	0.97 ***	1.54 ***
explained	0.01	0.90 ***	1.70 ***	1.75 ***		1.20 ***	0.63 ***	0.88 ***	0.86 ***	0.30	0.26 **	2.07 ***
median ratio HVA index	0.04	0.08	0.28 **	0.06		0.15 *	0.07 ***	0.13	-0.08	0.16	0.05 *	0.17 ***
owner	0.08	1.00 ***	1.27 ***	1.89 ***		1.19 ***	0.78 ***	0.91 ***	1.23 ***	0.33 ***	0.96 ***	2.04 ***
ihs(gifts & transfers)	0.04 *	0.00	0.00	-0.05 ***		-0.03 *		-0.03	-0.01	-0.33	-0.10 ***	0.01
years since largest transfer	-0.03	-0.01	-0.02	0.00		-0.02		0.01	0.01	0.06	0.00	0.00
Source: own calculations based on the HFCS UDB 1.0; results adjusted for multiple imputation; $* p<0.1 ** p<0.05 *** p<0.01$. All monetary values (total net wealth, total income and intergenerational transfers) are transformed using an inverse hyperbolic sine transformation in log form. For France, the macro index is used. Variables are grouped as follows: Employee status (self-employed, retired, other employment status, temporary contract, financial sector, public sector), demograph-	ed on the HF ional transfe oyee status (2	CS UDB 1.(rs) are tran. elf-employe); results ac sformed usi ed, unemple	ljusted for n ng an invers vyed, retire	ultiple imp se hyperboli d, other em	utation; * p ic sine trans ployment s	><0.1 ** p< sformation i tatus, tempe	0.05 *** p< 1 log form. ərary contre	:0.01. All mo For France, ıct, financia	JDB 1.0; results adjusted for multiple imputation; * p<0.1 ** p<0.05 *** p<0.01. All monetary values (total net wealth, tre transformed using an inverse hyperbolic sine transformation in log form. For France, the macro index is used. Varia employed, unemployed, retired, other employment status, temporary contract, financial sector, public sector), demogr	s (total net w ndex is used. lic sector), d	ealth, Variables emograph-
ics (male, age, age2, single, divorced, widowed, hhsize, mideduc, higheduc, born in the country of residence). Household total net wealth differences are relative to Germany. Standard errors are not bootstrapped.	divorced, w e not bootstra	idowed, hh pped.	size, midedı	uc, higheduc	c, born in th	e country o	f residence)	. Household	l total net we	alth differen	ces are relat	ive to

		Га	Table 19. Mil-OD John Percenture decomposition - Different property price specifications -	fferent pro	Different property price specifications	ce specifi	cations -						
		(1)	(2)	(3)	(4)	(2)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
		AT	BE	CY	ES	FR	GR	IT	ΓΩ	MT	NL	ΡT	SK
Ч 109	<u>Mean HVA index</u>	0 37 ***	1 38 ***	1 56 ***	1 33 ***		0 87 ***	1 76 ***	*** UU C	1 48 ***	0 75 ***	4** 3V U	*** US U
guo		0.18 *	0.54 ***	0.64 ***	0.22 ***		0.36 ***	0.37 ***	0.64 ***	0.35 ***	0.41 ***	0.03	0.40 ***
itiso To-F	mean HVA index	0.11 **	0.29 ***	0.16 ***	0.07 ***		0.07 *	0.06 ***	0.34 ***	0.17 ***	0.24 ***	0.04 **	0.21 ***
odu	owner	0.11 **	0.25 ***	0.30 ***	0.31 ***		0.35 ***	0.41 ***	0.24 ***	0.30 ***	0.22 ***	0.40 ***	0.23 ***
	ihs(gifts & transfers)	0.04 **	-0.01	0.00	-0.03 ***		-0.03 ***		-0.02 *	-0.01	-0.11	-0.06 ***	0.01 *
D	years since largest transfer	-0.03 **	0.00	-0.01	0.00 *		-0.02 ***		0.00	0.01 *	0.05	0.01	0.00
	<u>Macro index</u>												
ottie ottie	difference	-0.16 * 0.27 ***	1.44 *** 0 54 ***	0.63 ***	1.36 *** 0 28 ***	0.68 ***	0.37 ***	1.37 ***	2.02 *** 0.66 ***	1.47 *** 0 35 ***	0.78 ***	0.27 ***	-0.26 ***
	macroindev	0.00	0.04	-0.01	0.11 ***	0.21 ***	0.0	0.00 ***	0.00	0.00	0.30 ***	0.00	0.10 *
		0.00	····	10.0	TT-0	11.0		10.0	0.1.0 ***	11.0		*** 000	
	owner	-0.19 ***			0.32 ***	0.32 ***	0.25 ***	0.46 ***	0.35 ***	0.29 ***		0.33 ***	
	ihs(gifts & transfers)	0.01	-0.01	-0.11 **	-0.03 ***	0.03 ***	-0.10 ***		-0.02 *	-0.02 *	-0.07	-0.08 ***	-0.01
Я	years since largest transfer	0.00	0.00	** 60.0	0.01 **	-0.01 *	0.10 ***		0.00	0.02 **	0.03	0.00	-0.02
u	<u>Median HVA index</u>												
	difference	0.32 ***	1.38 ***	1.56 ***	1.33 ***		0.82 ***	1.26 ***	2.00 ***	1.48 ***	0.75 ***	0.45 ***	0.30 ***
	explained	0.19 *	0.54 ***	0.64 ***	0.21 ***		0.34 ***	0.36 ***	0.65 ***	0.35 ***	0.41 ***	0.03	0.41 ***
ore Star	median HVA index	0.11 **	0.27 ***	0.14 ***	0.04 ***		0.02	0.04 ***	0.36 ***	0.16 ***	0.23 ***	0.01	0.20 ***
	owner	0.11 **	0.27 ***	0.31 ***	0.32 ***		0.37 ***	0.42 ***	0.23 ***	0.31 ***	0.23 ***	0.42 ***	0.24 ***
1 Т 1 Т 1	ihs(gifts & transfers)	0.04 **	-0.01	0.00	-0.03 ***		-0.03 ***		-0.02 *	-0.01	-0.11	-0.06 ***	0.01 *
В	years since largest transfer	-0.03 **	0.00	-0.01	* 00.0		-0.02 **		0.00	0.01 *	0.05	0.01	0.00
u	<u>Median ratio HVA index</u>												
	difference	0.32 ***	1.38 ***	1.56 ***	1.33 ***		0.82 ***	1.26 ***	2.00 ***	1.48 ***	0.75 ***	0.45 ***	0.30 ***
	explained	0.16 *	0.53 ***	0.63 ***	0.21 ***		0.35 ***	0.36 ***	0.64 ***	0.34 ***	0.42 ***	0.03	0.40 ***
	median ratio index	0.08 **	0.25 ***	0.13 ***	0.04 ***		0.04	0.04 ***	0.33 ***	0.13 **	0.24 ***	0.01	0.19 ***
д Ч рээс	owner	0.11 **	0.28 ***	0.32 ***	0.32 ***		0.36 ***	0.42 ***	0.25 ***	0.33 ***	0.23 ***	0.42 ***	0.24 ***
	ihs(gifts & transfers)	0.04 **	-0.01	0.00	-0.03 ***		-0.03 ***		-0.02 *	-0.01	-0.11	-0.06 ***	0.01 *
Я	years since largest transfer	-0.03 **	0.00	-0.01	0.00 *		-0.02 **		0.00	0.01 *	0.05	0.01	0.00
Source: own calculation total income and inter- status (self-employed,	Source: own calculations based on the HFCS UDB 1.0; results adjusted for multiple imputation; * p<0.1 ** p<0.05 *** p<0.01. All monetary values (total net wealth, total income and intergenerational transfers) are transformed using an inverse hyperbolic sine transformation in log form. Variables are grouped as follows: Employee status (self-employed, memployed, retired, other employment status, temporary contract, financial sector, public sector), demographics (male, age, age2, single, di-	CS UDB 1.0; s) are transfo other empl	UDB 1.0; results adjusted for multiple imputation; * $p<0.1$ ** $p<0.05$ *** $p<0.01$. All monetary values (total net wealth, re transformed using an inverse hyperbolic sine transformation in log form. Variables are grouped as follows: Employee there employment status, temporary contract, financial sector, public sector), demographics (male, age, age2, single, di-	iusted for 3 an inver: tus, tempc	multiple in se hyperbo rary contr	nputation; lic sine tr act, finan	* p<0.1 * msformati cial sector	** p<0.05 on in log J ; public se	*** p<0.6 60rm. Var. ector), dem	1. All mor iables are ographics	tetary valu grouped a (male, age	es (total n s follows: ² , age2, s	net wealth, : Employee single, di-
vorced, widd bootstranned	vorced, widowed, hhsize, mideduc, higheduc, born in the country of residence). Household total net wealth differences are relative to Germany. Standard errors are not	uc, born in tl	ie country	of residenc	e). House	hold total	net wealth	difference	is are relai	tive to Ger	many. Star	udard erro	rs are not

$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$				- 22		ההביות היי	ne operili	cations -						
			(1)		(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
Addition Mode HVA index 0.21 0.66 0.21 0.66 0.21 0.66 0.21 0.66 0.21 0.66 0.21 0.66 0.21 0.66 0.21 0.22 0.21 0.22 <t< td=""><td></td><td></td><td>AT</td><td>BE</td><td>CY</td><td>ES</td><td>FR</td><td>GR</td><td>IT</td><td>ΓΩ</td><td>MT</td><td>NL</td><td>ΡT</td><td>SK</td></t<>			AT	BE	CY	ES	FR	GR	IT	ΓΩ	MT	NL	ΡT	SK
	ų	<u>Mean HVA index</u>												
Rest 0.06 0.21 0.06 0.01 0.15 0.00 0.11 0.06 0.11 0.06 0.01 0.15 0.00 0.11 0.06 0.01 0.11 0.05 0.04 0.06 0.01 <t< td=""><td></td><td>difference</td><td>0.21 ***</td><td>0.69 ***</td><td>1.03 ***</td><td></td><td></td><td>0.04</td><td>0.46 ***</td><td>1.23 ***</td><td>0.68 ***</td><td>0.24 ***</td><td>-0.21 ***</td><td>-0.64 ***</td></t<>		difference	0.21 ***	0.69 ***	1.03 ***			0.04	0.46 ***	1.23 ***	0.68 ***	0.24 ***	-0.21 ***	-0.64 ***
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	uoi	explained	0.08	0.24 ***	0.31 ***	0.00		0.12 ***	0.06 ***	0.28 ***	0.04	0.11	-0.23 ***	0.10
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	tisc	mean HVA index	0.01	0.15 ***	0.07	0.06 ***		0.05	0.07 ***	0.19 ***	0.11 *	0.24 ***	0.00	0.17 ***
Implementation Implementation 0.01 0.00 0.03 0.01	odu	owner	0.04 **	0.06 **	0.17 ***	0.12 ***		0.14 ***	0.14 ***	0.03	0.11 **	0.06 **	0.20 ***	0.09 ***
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		ihs(amount gifts & transfers)	0.02 **	-0.01	0.00	-0.03 ***		-0.02 **		-0.03 **	-0.02	-0.21 ***	-0.04 ***	0.01
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	D	years since largest transfer	-0.01	0.00	-0.01	0.00		-0.01		0.01	0.00	0.09	0.00	0.00
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	ųю	<u>Macro index</u>												
eq:composition fights \$\$ transfers) \$\$ 0.15 ** 0.15 ** 0.11 *** 0.01 \$\$ 0.11 *** 0.05 \$\$ 0.06 *** 0.32 *** 0.07 \$\$ 0.13 ** 0.23 *** 0.06 *** 0.14 *** 0.10 *** 0.11 *** 0.11			0.02	0.73 ***	0.34 **		0.31 ***		0.54 ***	1.27 ***	0.68 ***	0.25 ***	-0.31 ***	-1.01 ***
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	uoi		-0.15 **	0.24 ***	0.01	0.03	0.11 ***	0.05	0.06 ***	0.32 ***	0.07	0.13	-0.29 ***	0.03
Perform Oute Out Ou	tiso		-0.01	0.13 ***	-0.01	0.10 ***	0.14 ***	0.10 **	0.04 ***	0.17 ***	0.13 *	0.23 ***	0.06 **	0.10
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	odu		-0.11 ***	0.09 ***	-0.04	0.10 ***	0.05 ***	0.06 ***	0.17 ***	0.06 *	0.11 *	0.09 ***	0.14 ***	0.08
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$			0.01	-0.01	-0.10 **	-0.03 ***	0.02 ***	-0.08 ***		-0.03 **	-0.02 *	-0.20 ***	-0.06 ***	-0.01
	De	years since largest transfer	-0.01	0.00	0.03	0.01	-0.01 ***	0.06 **		0.01	0.01	0.08	0.00	0.05
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	ųı	<u>Median HVA index</u>												
Image: condition of the conditing the condition of the condition of the condition of	SZ	difference	0.21 ***	0.69 ***	1.03 ***	0.51 ***		0.04	0.46 ***	1.23 ***	0.68 ***	0.24 ***	-0.21 ***	-0.64 ***
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	uoi		* 60.0	0.23 ***	0.31 ***	-0.01		0.11 ***	0.05 **	0.29 ***	0.04	0.11	-0.23 ***	0.11
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	tisc		0.03	0.14 ***	0.06	0.04 ***		0.01	0.05 ***	0.20 ***		0.23 ***	-0.01	0.17 ***
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	odw		0.04 **	0.07 ***	0.17 ***	0.13 ***		0.16 ***	0.15 ***	0.03	0.11 **	0.07 ***	0.21 ***	0.10 ***
D years since largest transfer -0.01 0.00 -0.01 0.01 0.09 0.09 0.09 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.00 0.09 0.01 <t< td=""><td></td><td></td><td>0.02 **</td><td>-0.01</td><td>0.00</td><td>-0.03 ***</td><td></td><td></td><td></td><td>-0.03 **</td><td>-0.02</td><td>-0.21 ***</td><td>-0.04 ***</td><td>0.01</td></t<>			0.02 **	-0.01	0.00	-0.03 ***				-0.03 **	-0.02	-0.21 ***	-0.04 ***	0.01
Median ratio HVA index Median ratio HVA index 01 difference 0.21 *** 0.69 *** 1.03 *** 0.51 *** 0.046 *** 1.23 *** 0.68 *** 0.21 *** 0.21 *** 01 0.07 0.23 *** 0.31 *** 0.01 0.12 *** 0.04 0.46 *** 1.23 *** 0.24 *** 0.21 *** 02 explained 0.07 0.23 *** 0.31 *** 0.01 0.12 *** 0.02 ** 0.24 *** 0.21 *** 0.23 *** 0.24 *** 0.21 *** 0.23 *** 0.24 *** 0.21 *** 0.24 *** 0.21 *** 0.3 *** 0.21 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** <t< td=""><td>D</td><td>years since largest transfer</td><td>-0.01</td><td>0.00</td><td>-0.01</td><td>0.00</td><td></td><td>-0.01</td><td></td><td>0.01</td><td>0.00</td><td>0.09</td><td>0.00</td><td>0.00</td></t<>	D	years since largest transfer	-0.01	0.00	-0.01	0.00		-0.01		0.01	0.00	0.09	0.00	0.00
Difference 0.21 *** 0.69 *** 1.03 *** 0.51 *** 0.24 *** 0.24 *** 0.21 *** 0.01 0.21 *** 0.01 *** 0.21 ***	Чì	<u>Median ratio HVA index</u>												
Old explained 0.07 0.23 *** 0.31 *** -0.01 0.12 *** 0.05 *** 0.28 *** 0.04 0.12 -0.23 *** Perform median ratio HVA index 0.01 0.13 *** 0.05 0.04 *** 0.12 -0.23 *** 0.04 0.12 -0.23 *** Perform median ratio HVA index 0.01 0.13 *** 0.05 0.04 *** 0.09 0.24 *** -0.01 Perform owner 0.04 ** 0.08 *** 0.13 *** 0.13 *** 0.15 *** 0.03 0.12 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.04 *** 0.14 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 **** 0.01 *** 0.01 *** <td< td=""><td></td><td>difference</td><td>0.21 ***</td><td>0.69 ***</td><td>1.03 ***</td><td>0.51 ***</td><td></td><td>0.04</td><td>0.46 ***</td><td>1.23 ***</td><td>0.68 ***</td><td>0.24 ***</td><td>-0.21 ***</td><td>-0.64 ***</td></td<>		difference	0.21 ***	0.69 ***	1.03 ***	0.51 ***		0.04	0.46 ***	1.23 ***	0.68 ***	0.24 ***	-0.21 ***	-0.64 ***
Decision median ratio HVA index 0.01 0.13 *** 0.05 0.04 *** 0.03 0.05 *** 0.18 *** 0.09 0.24 *** 0.01 Proposition owner 0.04 ** 0.08 *** 0.13 *** 0.13 *** 0.03 0.05 *** 0.18 *** 0.09 0.24 *** 0.01 Proposition owner 0.04 ** 0.08 *** 0.18 *** 0.13 *** 0.01 0.03 0.12 ** 0.07 *** 0.21 *** 0.21 *** 0.21 *** 0.04 *** 0.21 *** 0.04 *** 0.44 *** 0.04 *** 0.16 *** 0.11 ** 0.07 ** 0.07 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.44 *** D years since largest transfer -0.01 0.00 -0.01 0.00 0.	uoi	explained	0.07	0.23 ***	0.31 ***	-0.01		0.12 ***	0.05 **	0.28 ***	0.04	0.12	-0.23 ***	0.10
Def product 0.04 ** 0.08 *** 0.13 *** 0.15 *** 0.03 0.12 ** 0.07 *** 0.21 *** 0.21 *** 0.04 *** 0.01 *** 0.03 0.12 ** 0.07 *** 0.21 *** 0.04 *** 0.01 *** 0.03 *** 0.01 *** 0.03 *** 0.04 *** 0.04 *** 0.01 *** 0.01 *** 0.01 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.04 *** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 *** 0.04 *** 0.01 *** 0.01 **	tiso	median ratio HVA index	0.01	0.13 ***	0.05	0.04 ***		0.03	0.05 ***	0.18 ***	0.09	0.24 ***	-0.01	0.16 ***
ihs(amount gifts & transfers) 0.02 ** -0.01 0.00 -0.03 *** -0.01 ** -0.01 ** -0.02 -0.21 *** -0.04 *** years since largest transfer -0.01 0.00 -0.01 0.00 -0.01 0.00 0.09 0.00	dw	owner	0.04 **	0.08 ***	0.18 ***	0.13 ***		0.16 ***	0.15 ***	0.03	0.12 **	0.07 ***	0.21 ***	0.11 ***
years since largest transfer -0.01 0.00 -0.01 0.00 -0.00 0.09 0.00		ihs(amount gifts & transfers)	0.02 **	-0.01	0.00	-0.03 ***		-0.01 **		-0.03 **	-0.02	-0.21 ***	-0.04 ***	0.01
	D	years since largest transfer	-0.01		-0.01	0.00		-0.01		0.01	0.00	0.09	0.00	0.00
	vorced, widc bootstrapped	vorced, widowed, hhsize, mideduc, higheduc, bootstrapped.		born in the country of residence). Household total net wealth differences are relative to Germany. Standard errors are not	of residenc	ce). House	hold total	net wealth	difference	s are rela	ive to Ger	many. Sta	ndard errc	rs are not

$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$				10 T	- Different property price specifications -	fferent pr	Different property price specifications	ice specifi	cations -						
				(1)		(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
				\mathbf{AT}	BE	CY	ES	FR	GR	IT	ΓΩ	МТ	NL	ΡT	SK
6 0.00 0.01 0	Ч ‡(ΣI;	ean HVA index												
Rest right expansion 0.00 0.01 0.11 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.01 0.00 0.01 0.00 0.01 0.00 0.01 0.00 0.01 0.00 0.01 0.01 0.00 0.01 0.00 0.01 0.00 0.01 0.00 0.01 0.00 0.01 0.01 0.00 0.01 0.00 0.01 0.01 0.00 0.01 0.00 0.01 0.01 0.00 0.01			ifference	0.22 ***	0.45 ***	1.11 ***	0.35 ***		-0.19 ***		1.08 ***	0.47 ***		-0.38 ***	-0.98 ***
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	юi		kplained	* 60.0	0.07 *	0.14	-0.11 **		-0.01	-0.01	0.19 **	-0.07	-0.18 *	-0.43 ***	0.13
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	tisc	uəc	mean HVA index	-0.02	0.01	0.12	0.04 *		0.00	0.06 ***	0.11 *	0.09	0.11 ***	0.00	-0.09
Ref Indemont glits & transfers) 0.02 0.00 0.01 <t< td=""><td>odu</td><td>or9^c</td><td>owner</td><td>0.03</td><td>0.05 *</td><td>0.06</td><td>0.07 ***</td><td></td><td>0.09 ***</td><td>0.08 ***</td><td>0.03</td><td>0.02</td><td>0.04</td><td>0.12 ***</td><td>0.26 *</td></t<>	odu	or9 ^c	owner	0.03	0.05 *	0.06	0.07 ***		0.09 ***	0.08 ***	0.03	0.02	0.04	0.12 ***	0.26 *
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		I	ihs(amount gifts & transfers)	0.02 *	0.00	0.00	-0.04 ***		-0.01		-0.02	-0.02	-0.28 ***	-0.04 ***	0.00
	Dé		years since largest transfer	0.00	0.00	-0.02	0.00		0.00		0.01	-0.01	0.04	0.00	-0.01
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	ų	Μ	acro index												
			ifference	0.15 **	0.50 ***	0.50 ***		0.17 ***			1.12 ***	0.50	0.03	-0.39 ***	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	uoj		vplained	0.00	0.08 *	0.08	-0.07	0.01	-0.02	-0.01	0.21 **	0.05	-0.15	-0.38 ***	-0.25
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	dise	ļuə:	macro index	0.00	0.02	0.00	0.11 ***	0.06 ***	0.08 **	0.04 ***	0.07	0.18 *	0.15 ***	0.09 ***	0.20 ***
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	odu	orə	owner	-0.07 **	0.05 **	0.05	0.04	0.03 ***	0.00	0.10 ***	0.06	-0.02	0.04 *	0.04	-0.11
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		I	ihs(amount gifts & transfers)	0.01	0.00	-0.06	-0.05 ***	0.02 ***	-0.05 **		-0.02	-0.02	-0.27 ***	-0.05 ***	-0.01
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	De		years since largest transfer	-0.01	0.00	-0.02	0.01	-0.01 ***	0.01		0.00	-0.02	0.05	0.01 *	0.05
$ \begin{array}{llllllllllllllllllllllllllllllllllll$	Чì	Μ	edian HVA index												
$ \begin{tabular}{l l l l l l l l l l l l l l l l l l l $	06	ц.	ifference	0.22 ***	0.45 ***	1.11 ***	0.35 ***		-0.19 ***	0.27 ***	1.08 ***	0.47 ***	-0.01	-0.38 ***	-0.98 ***
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	uoi		<pre></pre>	0.10 **	0.07 *	0.13	-0.12 **		-0.02	-0.02	0.19 **	-0.06	-0.18 *	-0.43 ***	-0.15 *
$ \begin{array}{ c c c c c c c c c c c c c c c c c c $	tisc	uəc	median HVA index	-0.01	0.00	0.11	0.02		-0.03	0.04 ***	0.12 *	0.10	0.11 ***	-0.01	0.09 **
$ \begin{array}{ c c c c c c c c c c c c c c c c c c $	dw	rer	owner	0.02	0.06 **	0.06	0.08 ***		0.10 ***	0.09 ***	0.03	0.01	0.04 *	0.12 ***	0.06
D years since largest transfer -0.01 0.00 -0.01 -0.01 0.04 0.00 Median ratio HVA index Median ratio HVA index 0.02 0.01 0.01 -0.01 0.04 0.00 Median ratio HVA index 0.22 0.45 0.11 *** 0.35 *** -0.19 *** 0.27 *** 0.01 0.03 *** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.44 *** 0.03 **** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.44 *** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.44 *** 0.43 **** 0.43 **** 0.43 **** 0.43 **** 0.43 <td< td=""><td></td><td>[</td><td>ihs(amount gifts & transfers)</td><td>0.02 *</td><td>0.00</td><td>0.00</td><td>-0.04 ***</td><td></td><td>-0.01</td><td></td><td>-0.02</td><td>-0.02</td><td>-0.28 ***</td><td>-0.04 ***</td><td>0.01</td></td<>		[ihs(amount gifts & transfers)	0.02 *	0.00	0.00	-0.04 ***		-0.01		-0.02	-0.02	-0.28 ***	-0.04 ***	0.01
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	D		years since largest transfer	-0.01	0.00	-0.02	0.00		0.00		0.01	-0.01	0.04	0.00	0.00
90 difference 0.22 *** 0.45 *** 1.11 *** 0.35 *** 0.19 ** 0.47 *** 0.01 0.38 *** 0.47 *** 0.01 0.38 *** 0.01 0.01 0.03 ** 0.11 ** 0.01 0.03 ** 0.11 ** 0.01 0.03 ** 0.11 ** 0.02 0.02 0.01 0.01 ** 0.13 * 0.47 ** 0.01 0.38 ** 0.00 wheth 0.01 0.01 0.11 ** 0.02 0.01 ** 0.03 * 0.03 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0.01 ** 0	ųı	M	edian ratio HVA index												
Diage explained 0.09 ** 0.07 * 0.13 -0.11 ** -0.02 -0.02 0.19 ** -0.07 -0.18 * -0.43 *** -0.43 *** Diage median ratio HVA index -0.02 0.00 0.11 0.02 -0.03 0.011 *** -0.01 Pic owner -0.03 * 0.06 ** 0.07 0.08 *** 0.09 *** 0.04 *** 0.01 *** -0.01 Pic owner 0.03 * 0.06 ** 0.07 0.08 *** 0.09 *** 0.04 * 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 *** 0.01 **** 0.04 **** 0.04 **** 0.04 **** 0.04 ***			ifference	0.22 ***	0.45 ***	1.11 ***	0.35 ***		-0.19 ***	0.27 ***	1.08 ***	0.47 ***		-0.38 ***	-0.98 ***
Description median ratio HVA index -0.02 0.04 *** 0.11 * 0.07 0.11 *** -0.01 Proposition owner 0.03 0.06 ** 0.07 0.01 *** **** **** 0.01	uoŗ		vplained	0.09 **	0.07 *	0.13	-0.11 **		-0.02	-0.02	0.19 **	-0.07	-0.18 *	-0.43 ***	-0.15 *
Def Def Owner 0.03 * 0.06 ** 0.07 0.08 *** 0.09 *** 0.04 0.03 0.04 * 0.12 *** 0.12 *** 0.04 * 0.12 *** 0.04 ** 0.01 **	tiso	uəɔ	median ratio HVA index	-0.02	0.00	0.10	0.02		-0.02	0.04 ***	0.11 *	0.07	0.11 ***	-0.01	0.09 **
ihs(amount gifts & transfers) 0.02 * 0.00 0.00 -0.04 *** -0.01 -0.02 -0.02 -0.28 *** -0.04 *** years since largest transfer 0.00 0.00 -0.02 0.00 0.00 0.00 0.01 -0.01 0.04 0.00	du	rer	owner	0.03 *	0.06 **	0.07	0.08 ***		0.09 ***	0.09 ***	0.04	0.03	0.04 *	0.12 ***	0.06
years since largest transfer 0.00 0.00 -0.02 0.00 0.00 0.00 0.00 0.0		-	ihs(amount gifts & transfers)	0.02 *	0.00	0.00	-0.04 ***		-0.01		-0.02	-0.02	-0.28 ***	-0.04 ***	0.01
	D		years since largest transfer	0.00	0.00	-0.02	0.00		0.00		0.01	-0.01	0.04	0.00	0.00
A VIA VIA VIA VIA VIA VIA VIA VIA VIA VI	vorcea, wiac bootstranned	widow	vorcea, widowea, nnsize, mideauc, nigneauc,		e country	of residen	ce). House	chota total	net weatth	attherence	S are rela	TIVE TO LIE	rmany Ma	NOOVO PYY	rs are not